

Contents lists available at ScienceDirect

Annual Reviews in Control



journal homepage: www.elsevier.com/locate/arcontrol

Review article

Motorway traffic flow modelling, estimation and control with vehicle automation and communication systems



Ioannis Papamichail^{*}, Nikolaos Bekiaris-Liberis¹, Anargiros I. Delis, Diamantis Manolis, Kiriakos-Simon Mountakis, Ioannis K. Nikolos, Claudio Roncoli², Markos Papageorgiou

School of Production Engineering and Management, Technical University of Crete, 73100 Chania, Greece

ARTICLE INFO

Article history: Received 22 July 2019 Accepted 9 September 2019 Available online 23 September 2019

Keywords: Vehicle automation and communication systems Traffic flow modelling Traffic flow estimation Traffic flow control

ABSTRACT

Traffic congestion on motorways is a serious threat for the economic and social life of modern society as well as for the environment, which calls for drastic and radical solutions. Conventional traffic management measures, currently applied, are valuable, but face limitations. During the last decades, a variety of vehicle automation and communication systems (VACS) have been developed and deployed, and many more are expected to appear in the near future. These systems provide a novel basis for a new generation of traffic management, which exploits emerging vehicle automation functions and connectivity channels to enable sensible traffic flow improvements in terms of efficiency and safety. A number of innovative concepts, tools and results that open up new horizons for traffic management research and practice in presence of VACS have been produced recently in the frame of TRAMAN21, an ERC Advanced Grant. This paper presents a collection of novel problems as well as of related traffic flow modelling, estimation and control developments for motorway traffic that can be used in the evolving traffic environment with VACS.

© 2019 Elsevier Ltd. All rights reserved.

Contents

1. 2.	Introduction Traffic flow modelling				
	2.1.	I. Microscopic modelling		327	
		2.1.1. Adaptive cruise control impact		327	
		2.1.2. Microscopic model calibration		328	
2.2. Macroscopic modelling			328		
		2.2.1. Second-order modelling in presence of VACS		329	
		2.2.2. Control-oriented motorway traffic modelling		331	
3.	Traff	affic state estimation using connected vehicle data		331	

[•] Corresponding author.

² Present address: Department of Built Environment, Aalto University, 00076 Aalto, Finland.

https://doi.org/10.1016/j.arcontrol.2019.09.002

1367-5788/© 2019 Elsevier Ltd. All rights reserved.

Abbreviations: ACC, Adaptive cruise control; CACC, Cooperative adaptive cruise control; CTM, Cell-transmission model; CV, Coefficient of variation; DE, Differential evolution; DP, Dynamic programming; ERC, European Research Council; FD, Fundamental diagram; FDA, Feasible direction algorithm; FIFO, First-in-first-out; GKT, Gas-kinetic traffic flow; IDM, Intelligent driver model; LCC, Lane changing control; LPV, Linear parameter varying; LQR, Linear quadratic regulator; MS, Merging sequence; MPC, Model predictive control; MTFC, Mainstream traffic flow control; NGSIM, Next Generation SIMulation; NLP, Nonlinear programming; PDE, Partial differential equation; PR, Penetration rate; QP, Quadratic programming; RM, Ramp-metering; TRAMAN21, TRAffic MANagement for the 21st century; TTT, Total travel time; TTS, Total time spent; UCO, Uniform complete observability; VACS, Vehicle automation and communication systems; V2I, Vehicle-to-infrastructure; V2V, Vehicle-to-vehicle; VMS, Variable message signs; VSL, Variable speed limit.

E-mail addresses: ipapa@dssl.tuc.gr (I. Papamichail), nikos.bekiaris@dssl.tuc.gr (N. Bekiaris-Liberis), adelis@science.tuc.gr (A.I. Delis), dmanolis@dssl.tuc.gr (D. Manolis), kmountakis@dssl.tuc.gr (K.-S. Mountakis), jnikolo@dpem.tuc.gr (I.K. Nikolos), claudio.roncoli@aalto.fi (C. Roncoli), markos@dssl.tuc.gr (M. Papageorgiou).

¹ Present address: School of Electrical and Computer Engineering, Technical University of Crete, 73100 Chania, Greece.

3.1. Cross-lane estimation				331			
		3.1.1.	Traffic density dynamics	331			
		3.1.2.	Estimation scheme	332			
		3.1.3.	Observability and stability	332			
		3.1.4.	Case studies	332			
	3.2.	3.2. Per-lane estimation		332			
		3.2.1.	Traffic density dynamics	332			
		3.2.2.	Estimation scheme	333			
		3.2.3.	Observability and stability	333			
		3.2.4.	Case studies	334			
4. Traffic flow control			ntrol	334			
	4.1. Cooperative merging						
	4.2. Vehicle trajectory control			336			
	4.3. Real-time ACC time-gap control			337			
	4.4. Lane-change control		ange control	338			
		4.4.1.	Introduction	338			
		4.4.2.	Linear multi-lane traffic flow model	338			
		4.4.3.	LQR problem formulation	339			
		4.4.4.	Stabilizability, detectability, and regulator	340			
		4.4.5.	Optimal set-point tuning via extremum seeking	340			
		4.4.6.	Policy-based density distribution at bottlenecks	340			
4.5. Integrated motorway network control		ed motorway network control	340				
		4.5.1.	Optimal control problem formulation	340			
		4.5.2.	Case study 1: open-loop optimization	342			
		4.5.3.	Case study 2: model predictive control in microscopic simulation	342			
_	4.6.	Other d	evelopments	342			
5.	Conc	lusions		343			
Funding.							
kererences							

1. Introduction

Traffic congestion on motorways is a serious threat for the economic and social life of modern society as well as for the environment, which calls for drastic and radical solutions. Some conventional traffic management measures currently applied may mitigate traffic congestion, but may also face various kinds of limitations. During the last decade, there has been a considerable effort to develop a variety of VACS that are expected to revolutionize the features and capabilities of individual vehicles within the next decades. Vehicles are goods manufactured and offered in a market, wherein they compete for the preference of the customers; therefore, most VACS are typically developed to benefit the individual vehicle and driver, often without a clear view or understanding for the implications, potential advantages and disadvantages they may have for the induced, accordingly modified traffic flow characteristics. From the community point of view, the gradual introduction of VACS brings along the necessity and continuously growing opportunities for accordingly adapted or utterly new traffic management actions and strategies aiming at a sensible decrease of traffic congestion and its detrimental implications for travel delays, traffic safety and the environment. It was the main objective of TRAMAN21 (www.traman21.tuc.gr), an ERC Advanced Grant, to develop foundations and first steps that pave the way towards a new era of future motorway traffic management research and practice, which is indispensable in order to accompany, complement and exploit the evolving VACS deployment.

VACS comprise on-board systems that undertake several vehicle functions at various levels of automation. Such systems, enhanced via novel communication features, aim at assisting or take on the driving task; but may also be exploited for improved traffic flow efficiency, see, Diakaki et al. (2014) and Diakaki, Papageorgiou, Papamichail and Nikolos (2015) for an overview. To this end, specific developments and options are required, particularly for network weak points, such as bottlenecks. In more detail, required contributions include the development of:

- microscopic and macroscopic traffic flow models and related simulation tools in presence of VACS;
- traffic state estimation algorithms and tools for the cross-lane case and for the per-lane case;
- traffic control algorithms and tools at various levels.

The main aim of this paper is to raise awareness and interest in novel problems and related approaches that emerge in various aspects of motorway traffic management due to the gradual introduction of VACS at increasing levels of sophistication and penetration. For each addressed area, the presentation outlines the respective motivation, essential issues and main pursued approaches in the technical literature (if any), before concentrating on the concise description of respective methods and results produced within the TRAMAN21 project. In this context, the three following sections address problems of traffic flow modelling, estimation and control, respectively.

Under the influence of vehicle automation (at different levels), the car-following and lane-changing vehicle behaviour is changing compared to manual driving. Such changes reflect also on the macroscopic characteristics of the emerging traffic flow. Section 2 discusses a microscopic simulation framework for ACCequipped vehicles and the calibration of a microscopic traffic simulator in presence of VACS for a real large-scale motorway network. Subsequently, a macroscopic multi-lane model with ACC and CACC vehicles at various penetration rates is presented, along with a novel numerical integration approach. Finally, a discrete (in space and time) first-order macroscopic traffic flow model, to be used in model-based optimization approaches, is also outlined.

In conventional traffic, real-time measurements are provided by spot sensors (based on different technologies), which are placed at specific motorway locations and monitor traffic variables at the respective specific locations. With the introduction of connected vehicles, which may transmit information to a traffic control centre, thus acting as mobile sensors, a new era of traffic monitoring and estimation has started. In this new context, Section 3 presents a novel real-time estimation scheme, for both cross-lane and lanebased traffic variable estimation, exploiting information from connected vehicles under virtually all penetration rates.

In the VACS era, a number of further significant changes emerge, which open up new prospects for traffic management and control:

- Connected vehicles may receive advice or commands from the traffic control centre, something that reduces, and potentially eliminates, the need for road-side actuators, such as traffic signals, VMS, VSL gantries etc.
- Vehicles may receive *individual* advice or commands, something that increases control granularity, compared to road-side actuators that convey the same message to all vehicles; this also enables new control actions, such as lane assignment and lane-change control, which are not feasible with conventional means.
- Vehicles may (be asked to) behave in a way that is beneficial for the whole traffic flow, i.e., they may become part of the traffic control logic; we call this possibility *vehicle-based traffic flow control*.

Section 4 presents a selection of new opportunities for more efficient traffic control enabled by VACS. First, some developments belonging to the exciting area of vehicle-based traffic flow control are presented, including: cooperative merging at motorway on-ramps; automated vehicle trajectory optimization; ACC-vehicle control for improved traffic flow; and feedback-based lane assignment. Subsequently, an integrated network-wide control approach is outlined, exploiting novel opportunities offered by VACS. Specifically, an optimization-based traffic control approach was developed, wherein the presence and use of VACS permits the implementation of an increased range of control actions in an integrated synergistic way. The obtained test results show significant improvements enabled by the combined use of VACS. Finally, a number of other applications are briefly touched upon. Section 5 concludes the overview of developed methods.

2. Traffic flow modelling

Since the penetration rate and sophistication level of VACS are still low, real data are sparse, and the main way to investigate their future impact on the traffic flow or to design, test and demonstrate novel traffic control approaches for various scenarios of VACS types, presence and use is modelling and simulation. This marks the very basic need of developing new or expanding existing traffic flow modelling and simulation methods and tools that account for the VACS presence. It must be emphasized that models are tools, which provide a necessary basis for a variety of different tasks, each requiring a different level of model granularity, accuracy or computational effort. Thus, in the following subsections, different traffic flow models (in presence of VACS) are presented, which serve as valuable tools for different purposes of motorway traffic management.

2.1. Microscopic modelling

Microscopic simulators are very useful and widely adopted tools for the analysis and management of transportation systems. However, when vehicles equipped with various kinds of VACS at various penetration levels are present in the traffic flow, there are no ready available simulation tools for immediate use. Different open-source (e.g., SUMO, see, Lopez et al., 2018) or commercial microscopic traffic simulators provide to the user the possibility to introduce self-developed car-following and lane-changing models, which reflect the modified behaviour of equipped vehicles. In the works reviewed in this paper, most reported microscopic simulation investigations are based on Aimsun (TSS-Transport Simulation Systems, 2014), which provides the necessary tools for introducing the VACS presence by coding self-developed vehicle models. Similar possibilities are also available with other commercial simulators, such as VISSIM, see Aria, Olstam and Schwietering (2016).

2.1.1. Adaptive cruise control impact

The control objectives of an ACC system are (Shladover, Su & Lu, 2012):

- *Speed control mode:* to travel close to the pre-set (by the driver) desired speed, in cases where no leading vehicle is detected by the vehicle's sensors; or a leading vehicle is detected but its speed is higher than the driver-defined desired speed.
- *Gap control mode:* to adjust the speed of the equipped vehicle to the speed of the leading vehicle and maintain the user-defined desired time-gap to the leading vehicle, in cases that a leading vehicle is identified by the sensors, and its speed is lower than the pre-set by the user desired speed.

Several works investigated the influence of ACC systems on traffic flow, attempting to capture the impact of different settings (mainly different time-gaps) and penetration rates, by use of microscopic simulation, see e.g., Arnaout and Bowling (2011), Bayar, Sajadi-Alamdari, Viti and Voos (2016), Schakel, Van Arem and Netten (2010) and Shladover et al. (2012). General conclusions that may be drawn from those studies are that: (i) ACC systems have the potential to improve or deteriorate, depending on their settings, the traffic conditions compared to the case of conventional manually driven vehicle traffic; and (ii) the level of the influence is closely related to the ACC penetration rate. In particular, Ntousakis, Nikolos and Papageorgiou (2015) reviewed previous modelling efforts reported in the literature, concerning ACC-equipped vehicles, and discussed some critical aspects to be considered when designing or simulating such systems. Moreover, a microscopic simulation framework for ACC-equipped vehicles was developed within the Aimsun microscopic traffic flow simulator, utilizing its so-called API and MicroSDK tools. Simulation experiments have been performed to examine:

- (a) the impact of ACC on traffic flow capacity for different penetration rates and different ACC time-gap settings for a single-lane open-stretch road; and
- (b) the effect of ACC on preventing the formation of stop-and-go waves in a ring-road.

For simulating manually driven vehicles, two different carfollowing models were employed, the one by Gipps (1981) and IDM (Treiber, Hennecke & Helbing, 2000). The simulation results for case (a) showed that the ACC time-gap has a direct impact on the capacity: the smaller the time-gap, the higher the capacity, and vice-versa. Thus, the traffic flow capacity can be increased if the ACC time-gap is smaller than that of manual vehicles; else deterioration of capacity may occur. Fig. 1 illustrates that, when ACC and manual vehicles have a similar time-gap (here around 1.1 s), then the ACC vehicles have no influence on capacity for any penetration rate. However, if the ACC time-gap is higher (lower) than that employed by manual vehicles, then any increase of the ACC penetration leads to a decrease (increase) of capacity; and the potential changes at both ends of the spectrum are indeed substantial. These results highlight the need and potential for appropriate use of ACC systems with regard to traffic management, as discussed in Section 4.3.

For the case (b), it was observed that ACC vehicles may improve the stability of traffic flow, since they mitigate string-instability



Fig. 1. Traffic flow capacity (veh/h/lane) for different ACC time-gaps and different ACC penetration rates.

and hence the amount and intensity of stop-and-go waves. Such observations were indeed made also in other similar works, indicating that ACC systems have the potential to smooth traffic flow, decrease fuel consumption, and improve traffic flow efficiency (loannou & Stefanovic, 2005; Ngoduy, 2012). However, it should be stressed that the real dynamic behaviour of real ACC vehicles is not well-known and hence not necessarily well-captured in microscopic simulation studies that are based on idealized ACC timegap regulators. In fact, some more recent studies (Gunter et al., 2019; Gunter, Janssen, Barbour, Stern & Work, 2019; Milanés & Shladover, 2014), employing data from real ACC vehicles, indicate that real ACC vehicles may behave (like manually-driven vehicles) in a string-unstable way. Corresponding model identification exercises for ACC vehicles using real data may enable more realistic assessment of the impact of ACC systems on the traffic flow.

2.1.2. Microscopic model calibration

Microscopic simulation is a valuable tool for assessment of novel control strategies that exploit VACS of various kinds. Although there is some uncertainty regarding the exact dynamic behaviour of equipped vehicles, a microscopic simulator should provably reflect with sufficient accuracy the traffic flow resulting when only manually driven vehicles are considered. Calibration and validation of microscopic models by use of real data for conventional traffic has been the subject of many recent studies, see e.g., Toledo, Ben-Akiva, Darda, Jha and Koutsopoulos (2007) and Treiber and Kesting (2012). In order to produce a simulationbased testbed for a variety of traffic control investigations, a thorough calibration of a microscopic multi-lane traffic flow model for a real sizable motorway was carried out by Perraki, Roncoli, Papamichail and Papageorgiou (2018). The modelled network is a 12 km-long stretch of the motorway A20, from Rotterdam to Gouda, The Netherlands (Fig. 2(a)). Its topological and traffic characteristics (it includes a lane-drop, strong merging flow from on-ramps and partly saturated off-ramps) make it a very interesting testbed for investigating various different scenarios with and without the presence of VACS. The microscopic simulator Aimsun was used, with critical additions and adaptations made in order to improve the realism of the simulation and enable it to reproduce complex traffic phenomena (e.g., the capacity drop at the head of congestion). More specifically, different new features were introduced, including the IDM car-following model (Treiber et al., 2000) as well as improved heuristic rules to capture more accurately the lane-changing behaviour in the proximity of on-ramps and lane-drops (vehicle merging). Fig. 2(b) illustrates the similarity of the real data and model outcome, respectively, in terms of mean speed. Note that the shock wave entering the stretch from the downstream boundary in Fig. 2(b-upper) is not reproduced by the model, which was intentionally not fed with the corresponding boundary conditions to avoid interference with congestion triggers that are not included in the considered stretch.

This model was eventually enriched with features related to VACS (e.g., connected vehicles, partially and fully automated cars) and was used for testing the accuracy of proposed macroscopic models (see Section 2.2), estimation tools (see Section 3), as well as various proposed control strategies (see Section 4).

2.2. Macroscopic modelling

Macroscopic models characterize traffic in terms of aggregate density, space-mean speed, and flow, as opposed to microscopic ones that describe interactions between individual vehicles Macroscopic traffic flow modelling has a long history, see (Hoogendoorn & Bovy, 2001) for a review. Macroscopic models are valuable for traffic management purposes, because, except for their use as efficient simulators (see e.g., Papageorgiou, Papamichail, Messmer & Wang, 2010), they are analytic in nature and may therefore provide a solid basis for the application of modern estimation and control tools, see, Kotsialos and Papageorgiou (2001). In the context of VACS, conventional macroscopic models need to be appropriately modified and extended. To start with, the modified microscopic behaviour of equipped vehicles, appearing in different penetrations, reflects into accordingly modified macroscopic traffic behaviour, which needs to be captured not merely via modified model parameters, but also via structural changes in the model equations. In addition, VACS enable utterly new control measures, such as lane-change commands to individual vehicles, which are not possible in conventional traffic. To support the design and testing of such lane-change control strategies, macroscopic models need to be lane-based, i.e., able to describe not only longitudinal flow, but also lateral (lane-changing) flow, something very rare in conventional traffic flow modelling.

Available conventional macroscopic models feature significant differences in complexity and potential accuracy. Macroscopic models may be first-order (comprising one PDE) or second-order (two PDEs) or higher-order. For the convenience and computational efficiency of their application, macroscopic models may be discrete in space and time, i.e., comprise a number of difference state equations instead of PDEs. Clearly, like in other domains, there is no best model form for any use; rather the most appropriate macroscopic model form depends on its intended use.



Fig. 2. (a) A real motorway testbed; (b) Space-time plot of mean speed (real data - upper row - and microscopic model - lower row).

The outlined variety of different conventional macroscopic models reflects also in their extension for the case of VACS presence and is probably a reason why macroscopic model developments including VACS are still scarce. In the following, a PDE-based second-order macroscopic multi-lane traffic flow model with ACC and CACC vehicles at various penetration rates is presented first. Then, a simpler discrete first-order multilane macroscopic traffic flow model, to be used in model-based optimization approaches is outlined.

2.2.1. Second-order modelling in presence of VACS

Essentially, macroscopic models are governed by the continuity equation, which is derived from vehicle conservation and describes the evolution of density based on flow gradients. In addition, second-order models include a second PDE to describe how speed changes dynamically. Some parameters from this second equation can be used to differentiate the characteristics between manual vehicles and those equipped with ACC or CACC systems. Although the literature following this modelling approach is rather limited, the approach has been used to model ACC (Wang & Rajamani, 2004; Yi & Horowitz, 2006), V2V communication (Ngoduy & Jia, 2017; Zheng, Jin & Huang, 2015) and CACC (Ngoduy, 2013, 2009). Although most of literature following this approach focuses on homogeneous traffic, i.e., assuming 100% penetration rate, there exist some very recent attempts that consider varying penetration rates of ACC or CACC systems (Ngoduy, 2012).

In this section, we present a PDE-based multi-lane GKT traffic flow model (Delis, Nikolos & Papageorgiou, 2014; Shvetsov & Helbing, 1999; Treiber & Kesting, 2013) extended with ACC or CACC modelling terms to account for the effect on traffic dynamics of variable penetration rates of ACC or CACC vehicles, as well as of different classes of drivers using different time-gaps. In a highway with *N* lanes, numbered by l = 1, 2, ..., N, with l = 1 being the slowest lane, we aim to model the evolution, in space *x* and time *t*, of traffic density (number of vehicles per unit length) $\rho_l(x, t)$ and traffic flow rate (number of vehicles per unit of time) $q_l(x, t) = \rho_l u_l$, with $u_l(x, t)$ being the mean speed of vehicles in the *l*th lane. To this end, the extended ACC/CACC multi-lane GKT model may be written for each lane *l* as

$$\partial_t \rho_l + \partial_x q_l = \sum_{k=1}^m \rho_{rmp,k} \delta_{j,1} + w_l^{(1)}$$
(1)

$$\partial_{t}q_{l} + \partial_{x}\left(q_{l}^{2}/\rho_{l} + \theta_{l}\rho_{l}\right) \\ = \sum_{k=1}^{m} h_{rmp,k}\delta_{l,1} + w_{l}^{(2)} + \left(\frac{\rho_{l}V_{e,l}^{*} - q_{l}}{\tau_{l}}\right) [1 - PF_{l}(\rho_{l})] + P\mathcal{V}_{(c)acc,l}$$
(2)

$$\rho_l(x,0) = \rho_{l,0}(x); q_l(x,0) = q_{l,0}(x), \quad l = 1, 2, \dots, N.$$
(3)

System (1)-(3) constitutes a weakly coupled system of conservation laws, with a total of 2*N* equations, for traffic density and flow for each lane. In what follows, definitions and explanations related to these equations are provided step-by-step:

The core multi-lane model: In the momentum (flow) Eq. (2), we introduce the modelling terms that account for the ACC or CACC vehicles in the flow, namely the terms $[1 - PF_l(\rho_l)]$ and $PV_{(c)acc, l}$. The parameter *P* in these terms accounts for the penetration rate of ACC or CACC vehicles. For *P* = 0, the original multi-lane GKT

model for manually driven vehicles is recovered. In the flux term of Eq. (2), $\theta_l \rho_l$ represents a pressure-like term, where $\theta_l = A_l(\rho_l)u_l^2$ with $A_l(\rho_l)$ being a density-dependent variance factor given by the Fermi function as $A_l(\rho_l) = A_{0,l} + \delta A_l [1 + \tanh((\rho_l - \rho_{cr,l})/\delta \rho_l)]$, where $\rho_{cr,l}$ is the critical density for the *l*th lane, with $A_{0,l}$ and $A_{0,l} + 2\delta A_l$ the variance pre-factors between the two states; while $\delta \rho_l$ denotes a transition density width. Typical range of values for the constants $A_{0,l}$, δA_l and $\delta \rho_l$ along with the typical range of the other parameters for this model can be found in Delis et al. (2014), Delis, Anargiros, Nikolos and Papageorgiou (2018), 2015a and Ngoduy (2013). Term $\rho_{rmp,k}$ ($h_{rmp,k}$) models the incoming (outgoing) traffic flow from the kthon-ramp (or to the kthoff-ramp) on the first lane.

The model also includes a (non-local) relaxation term in the flow Eq. (2) aiming to keep flow in equilibrium at each lane. We denote with $V_{e,l}^* \equiv V_e^*(\rho_l, u_l, \rho_{\alpha,l}, u_{\alpha,l})$ the non-local and dynamic equilibrium speed, with τ_l being a relaxation time. $V_{e,l}^*$ depends not only on the local lane density ρ_l and mean speed u_l , but also on a non-local density $\rho_{a,l}$ and mean speed $u_{a,l}$, and is defined as

$$V_{e,l}^* = u_{\max,l} \left[1 - \frac{\theta_l + \theta_{\alpha,l}}{2A_l(\rho_{\max,l})} \left(\frac{\rho_{\alpha,l}T_l}{1 - \rho_{\alpha,l}/\rho_{\max,l}} \right)^2 B(\delta u_l) \right].$$
(4)

 $V_{e,l}^*$ is given by the maximum lane velocity $u_{\max, l}$, reduced by a term that reflects necessary deceleration maneuvers. Both $\rho_{\alpha, l}$ and $u_{\alpha, l}$ are computed at an anticipated location $x_{\alpha, l} = x_l + \gamma_l(1/\rho_{\max, l} + T_l u_l)$ with T_l being the desired time-gap for manual driving, $\rho_{\max, l}$ the maximum lane density and γ_l a scale factor. Finally, *B* is a so-called Boltzmann (interaction) factor, which, with $\delta u_l = (u_l - u_{\alpha, l})/(\sqrt{\theta_l - \theta_{\alpha, l}})$, is determined as

$$B(z) = 2 \left[z \frac{e^{-z^2/2}}{\sqrt{2\pi}} + \left(1 + z^2 \right) \int_{-\infty}^{z} \frac{e^{-y^2/2}}{\sqrt{2\pi}} dy \right].$$
 (5)

Multi-lane terms: The basic assumptions that the multi-lane model aims to satisfy, in terms of vehicles' interactions are: (a) changing to the left; (b) changing to the right; (c) breaking; (d) acceleration-following; (e) acceleration-free flow (Delis et al., 2018; Delis, Nikolos & Papageorgiou, 2015b). To this end, source terms $w_l^{(1),(2)}$ in the model equations account for the sources and sinks due to lane-changing for the quantities $v_l^{(1)} = \rho_l$ and $v_l^{(2)} = q_l$ that satisfy the above assumptions

$$w_{l}^{(1),(2)} = \left(\frac{1}{T_{l-1}^{L}}v_{l-1}^{(1),(2)} - \frac{1}{T_{l}^{R}}v_{l}^{(1),(2)}\right)\left(1 - \delta_{l,1}\right) \\ + \left(\frac{1}{T_{l+1}^{R}}v_{l+1}^{(1),(2)} - \frac{1}{T_{l}^{L}}v_{l}^{(1),(2)}\right)\left(1 - \delta_{l,N}\right)$$
(6)

where terms $1/T_l^L$ and $1/T_l^R$ are the lane changing rates from lane l to the left l+1 and right l-1 lane, respectively, and $\delta_{i,j}$ the Kronecker delta. These rates are given as

$$\frac{1}{T_l^L} = P_L(\rho_{l+1})\sigma(\rho_l) + S_l^L,$$

$$\frac{1}{T_l^R} = P_R(\rho_{l-1})(1 - P_L(\rho_{l+1}))\sigma(\rho_l) + S_l^R,$$
 (7)

with $P_{R, l}(\rho)$ being the lane-changing probabilities due to vehicle interactions and $\sigma(\rho) = \sigma_f (1 - \rho_l) \rho_l^2$ denote the interaction frequencies regarding decelerations and accelerations.

Furthermore in (7) we take into account spontaneous lane changes, which are not caused by interactions with other vehicles, through the terms $S_l^{L,R}$. We assume that these terms depend on the density, ρ_l , and are given as

$$S_{l}^{L,R} = k_{l}^{L,R} \left(1 - \frac{\rho_{l\pm 1}}{\rho_{\max,l\pm 1}} \right)^{\rho}$$
(8)

where $k_{i}^{L,R}$ and β are spontaneous lane-changing parameters.

The ACC/CACC terms: Terms $[1 - PF_l(\rho_l)]$ and $PV_{(c)acc, l}$ in the flow Eq. (2) are related to the extension of the multi-lane model, so as to incorporate the effects of ACC/CACC in traffic dynamics. The control objectives of an ACC/CACC system, mentioned earlier, should be satisfied here (Delis, Nikolos & Papageorgiou, 2016; Nikolos, Delis & Papageorgiou, 2015).

At densities below a threshold $\rho_{acc, l}$ (being lower than or equal to the critical lane density $\rho_{cr, l}$), the additional term to the multilane GKT model has no influence, since it is assumed that the vehicles apply the desired speeds as in manual driving. At densities around $\rho_{acc, l}$, a smooth but fast transition between the previous manual case and the ACC or CACC model is established, using the Fermi-like function $F_l(\rho_l) = 0.5[1 + \tanh((\rho_l - \rho_{acc, l})/\Delta\rho_l)]$ with $\Delta \rho_l \approx 0.025 \rho_{max, l}$. As a result, the corresponding source for ACC vehicles in each lane reads

$$V_{acc,l}(\rho_{l}, u_{l}, \rho_{l}^{*}, u_{l}^{*}) = F_{l}(\rho_{l}) \left(\frac{\rho_{l}^{*} u_{l}^{*} - \rho_{l} u_{l}}{\tau_{l}^{*}}\right)$$
(9)

which forces the speed to relax to the speed of the preceding vehicle u_l^* after a relaxation time τ_l^* . Assuming that different classes of drivers choose different desired time-gaps, we denote ω_j , $j = 1, \ldots, Nclass$, the percentage of drivers with different time-gaps T_j^* . These gaps can be imposed through their corresponding influence on a desired lane density ρ_l^* in (9), that is given as:

$$\rho_l^* = \frac{1}{\left(1/\rho_{max,l} + \sum_{j=1}^{Nclass} \omega_j T_j^* u_{j,l}^*\right)}$$
(10)

where $u_{j,l}^* \equiv u(x_{j,l}^*)$ is evaluated at location $x_{j,l}^* = x_l + \gamma_l^*(1/\rho_{max,l} + T_j^*u_l)$ and the mean preceding vehicle speed in (9) $u_l^* = \sum_{j=1}^{Nclass} \omega_j u_{j,l}^*$. Indicative values used for ACC traffic are $T^* \in [0.8, 2.2]$ s, following [ISO 15622, 2010] standards, and $\tau^* \approx 1$ s.

To model the impact of CACC vehicles in traffic flow dynamics, a similar approach is utilized, with the only difference being that a vehicle can exchange information with M preceding vehicles, each of which has a different relaxation time. Thus, the corresponding source term of the momentum equation for lane l takes now the form

$$V_{cacc,l}(\rho_l, u_l, \rho_l^*, u_l^*) = F_l(\rho_l) \sum_{i=1}^{M} \left(\frac{\rho_l^* u_l^* - \rho_l u_l}{\tau_l^*} \right)$$
(11)

where now

$$\rho_l^* = \frac{1}{\left(1/\rho_{max,l} + \sum_{j=1}^{Nclass} \omega_j T_j^* u_{j,1,l}^*\right)}, \qquad u_{i,l}^* = \sum_{j=1}^{Nclass} \omega_j u_{j,i,l}^*$$
(12)

with $u_{j,i,l}^* \equiv u_l(x_{j,i,l}^*)$ at location $x_{j,i,l}^* = x_l + i\gamma_l^*(1/\rho_{\max,l} + T_j^*u_l)$. For example, for CACC systems, one can assume that M = 3, while $[\tau_{1,l}^*, \tau_{2,l}^*, \tau_{3,l}^*] = [2, 3, 6]$ as in (Delis et al., 2015b, 2016; Delis et al., 2018).

The numerical integration of the PDE model equations is based on an accurate and robust higher-order finite-volume relaxation scheme. A fifth-order weighted essential non-oscillatory-type interpolant approach is used for the spatial discretization, while time integration is based on a third-order implicit-explicit Runge–Kutta method (Delis et al., 2014, 2015b).

The developed macroscopic model has been used to simulate traffic flow for either manual or ACC/CACC or mixed vehicle traffic, in single-lane or multi-lane highways (Delis et al., 2016, 2018; Nikolos et al., 2015; Porfyri, Delis, Nikolos & Papageorgiou, 2017, 2016; Strofylas, Porfyri, Nikolos, Delis & Papageorgiou, 2018). In the work by Delis et al. (2018), the multi-lane second-order GKT model, extended with the modelling of ACC/CACC vehicles was presented and evaluated. A real motorway network has been used

as a test-case, where a recurrent congestion occurs, and for which real traffic measurements are available for manual traffic. The main parameters of the multi-lane GKT model for manual driving were calibrated and validated; eventually, keeping these parameters fixed, simulations were conducted for various penetration rates of ACC and CACC vehicles, assuming a constant time-gap, which was different for each case. The numerical simulations demonstrate that ACC systems with low time-gap may mitigate congestion at bottlenecks at an extent that depends on their penetration rate; and that CACC systems lead to higher efficiency at the same penetration rates. In Strofylas et al. (2018), numerical simulations have been presented using real traffic data from a freeway stretch of Attiki Odos motorway, Greece, where recurrent congestion occurs during the morning peak hours. A parallel, synchronous or asynchronous, metamodel-assisted DE algorithm was employed for the calibration of the second-order macroscopic GKT model. Two Artificial Neural Networks, a multi-layer perceptron and a radial basis function network, were used as surrogate models to decrease the computation time of the evaluation phase of the DE optimizer. The obtained results demonstrate that the proposed model is reasonably accurate in reproducing traffic dynamics, while the parallel DE algorithm can be effectively used for its calibration.

2.2.2. Control-oriented motorway traffic modelling

As mentioned earlier, macroscopic models may serve different purposes, some of which require availability of simpler than PDE-based models to enable higher computational efficiency and methodological simplicity. Network-wide integrated traffic control strategies employing, beyond conventional ramp metering, appropriate VACS-based actuators, such as vehicle speed control and lane-assignment or lane-changing recommendations in the aim of mitigating motorway traffic congestion are presented in Section 4. To enable some of these approaches by use of optimal control methods, an appropriate low-complexity traffic flow model is needed, both for control strategy design and as a no-control base case for comparative evaluation studies.

Such a multi-lane first-order space-time discrete model was built starting from the well-known CTM (Daganzo, 1994), which was modified and extended to consider additional important aspects of the traffic dynamics, such as lane changing and the capacity drop. The model was derived with a view to combine realistic traffic flow description with a simple discrete piecewiselinear mathematical structure, which can be exploited for efficient optimal control problem formulations, as well as for the development of advanced model-based nonlinear feedback control concepts, as will be reported in Section 4. Although the model was primarily designed for use in future traffic conditions including VACS, it may also be employed, with appropriate parameter values, for conventional traffic flow representation. The accuracy of the simple model was demonstrated through calibration and validation procedures using conventional real data from an urban motorway located in Melbourne, Australia, see, Roncoli, Papageorgiou and Papamichail (2015b and Section 4.5 for more details.

Regarding, in particular, the crucial phenomenon of capacity drop, it is well-known that classic first-order traffic flow models are not able to reproduce it. Therefore, various research groups investigated different possibilities of including capacity drop in a first-order traffic flow model (Ferrara, Sacone & Siri, 2015; Li, Liu, Xu & Wang, 2016; Manolis, Papamichail, Kosmatopoulos & Papageorgiou, 2016; Muralidharan & Horowitz, 2015; Ngoduy, 2012; Torné, Soriguera & Geroliminis, 2014). To comprehensively address this fundamental issue, some modelling approaches presented in the literature, together with some novel approaches, were analysed, with particular emphasis on the practical applicability of such models for traffic management and control. A subset of the most promising modelling approaches was thoroughly tested, calibrated and compared using real data from a motorway network in U.K., and the obtained results, reported in (Kontorinaki, Spiliopoulou, Roncoli & Papageorgiou, 2017), may guide the proper choice of a simple traffic flow model for use in various traffic control endeavours.

3. Traffic state estimation using connected vehicle data

The availability of real-time data is a prerequisite for traffic management and control. Spot detectors are conventionally used, but they are costly to acquire, install and maintain; while the data they provide are local, rather than spatial, as required by traffic control strategies. Abundant information stemming from equipped vehicles may be used to reduce the cost and increase the accuracy of real-time information. In fact, with the introduction of VACS of various kinds, an increasing number of vehicles become "connected," i.e., enabled to send (and receive) real-time information to a local or central monitoring and control unit. Thus, connected vehicles may communicate their position, speed and other relevant information, i.e., they can act as mobile sensors. The transforming of such information into real-time estimates for the traffic states calls for appropriate estimation schemes, and various research works attempt to exploit these data for travel time or highway state estimation; employing various kinds of traffic models and estimation schemes, see e.g., Deng, Lei & Zhou (2013), Herrera and Bayen (2010), Piccoli, Han, Friesz, Yao and Tang (2015), Seo, Kusakabe and Asakura (2015) and Yuan, Van Lint, Van Wageningen-Kessels and Hoogendoorn (2014). These estimation approaches involve some sort of empirical traffic flow modelling, typically a fundamental diagram, which calls for appropriate model calibration and validation to the specific highway traffic conditions prior to deployment. An alternative estimation approach was developed by Bekiaris-Liberis, Roncoli and Papageorgiou (2016), which employs only exact traffic flow physical laws, such as the conservation-ofvehicles equation, which eliminates the need for model calibration prior to usage. The method exploits position and speed data from connected vehicles, as well as total flow measurements obtained from a minimum number of fixed detectors using a timevariant Kalman filter. Observability of the underlying model and stability of the estimators are rigorously addressed. Cross-lane and per-lane traffic state estimation schemes are developed and validated using both real data and data generated via microscopic simulation.

3.1. Cross-lane estimation

3.1.1. Traffic density dynamics

The discrete-time dynamics of total densities ρ_i of vehicles in *N* highway segments are given by the conservation equation

$$\rho_i(k+1) = \rho_i(k) + \frac{T}{\Delta_i}(q_{i-1}(k) - q_i(k) + r_i(k) - s_i(k)),$$
(13)

where, for all traffic variables, we denote by index *i* their value in segment *i*; q_i is the total flow; *T* is the time step, Δ_i is the length of segment *i*, and k = 0, 1, ... is the discrete time index. The variables r_i and s_i denote the unmeasured inflow and outflow of vehicles at on-ramps and off-ramps, respectively, whose dynamics may be reflected by a random walk. Thus, using the known relation

$$q_i = \rho_i \nu_i, \tag{14}$$



Fig. 3. Mean and standard deviation of the speed error (connected-vehicles versus all-vehicles speed) as the penetration rate of connected vehicles varies. (a) The case of NGSIM data; (b) The case of data generated in microscopic simulation.

where v_i is the mean speed in segment *i*, we get

$$\rho_i(k+1) = \frac{T}{\Delta_i} v_{i-1}(k) \rho_{i-1}(k) + \left(1 - \frac{T}{\Delta_i} v_i(k)\right) \rho_i(k) + \frac{T}{\Delta_i} \theta_i(k), \theta_i(k+1) = \theta_i(k),$$
(15)

with $\theta_i = r_i - s_i$. Assuming that the mean speed of all vehicles is not systematically different from the average speed of connected vehicles, denoted by v_i^c , and defining $x = (\rho_1, \dots, \rho_N, \theta_1, \dots, \theta_N)^T$, system (15) can be written in the form of a linear parameter-varying system of the form

$$x(k+1) = A(v^{c}(k))x(k) + Bu(k), y(k) = Cx(k),$$
(16)

for appropriate A, B, C where u is the measured mainstream inflow, and y consists of a minimum amount of measured flows, which are necessary to guarantee observability; all these flow measurements are obtained from fixed flow detectors.

3.1.2. Estimation scheme

For estimating vector *x*, we employ a Kalman filter (see e.g., Cha, Rotkowitz & Anderson, 2008) using model (16) and measurements v^c , *y*, *u*. As mentioned, a main assumption of the estimation approach is that the average of speed measurements stemming from connected vehicles reports is representative of the all-vehicles average speed. Fig. 3 validates this assumption, showing the noise statistics derived for the speed measurement error using both real data (specifically, the processed by Montanino and Punzo (2013) real microscopic traffic data collected within the NGSIM program, US DoT, 2005) as well as traffic data generated using microscopic simulations (namely, simulations in Aimsun see, Fountoulakis, Bekiaris-Liberis, Roncoli, Papamichail & Papageorgiou, 2017, for details).

3.1.3. Observability and stability

To guarantee the stability of the estimation scheme Bekiaris-Liberis et al., 2016), the most crucial element is establishing the UCO property of system ((16). Assuming that the mean speeds are nonzero for all times, it can be shown, using an algebraic or a graph-theoretic approach Bekiaris-Liberis et al., 2016; Bekiaris-Liberis, Roncoli & Papageorgiou, 2017), that ((16) is UCO when a fixed flow detector is placed, in addition to the highway mainstream exit and entry, at every segment immediately upstream a ramp whose flow is unmeasured, as illustrated in Fig. 4(a). In practice, observability is guaranteed when a fixed flow detector is placed anywhere between two consecutive unmeasured ramps (Bekiaris-Liberis et al., 2017).

3.1.4. Case studies

Real data validation using NGSIM data: We employ NGSIM traffic data (see Section 3.1.2) for a 400-m stretch of highway I-80, Emeryville, California. The stretch is divided into 8 homogeneous segments of 50 m in length, and an on-ramp is located within the fourth segment. All necessary information needed for estimation is extracted from the available vehicle trajectory data, i.e., v^{c} (computed as average of speeds of a subset of vehicles randomly tagged as connected), *y*, *u*. For brevity, only results for 5% penetration rate are shown in Fig. 5; see Roncoli, Bekiaris-Liberis and Papageorgiou (2016b) for details.

Microscopic simulation-based validation: For various percentages of connected vehicles, we use as evaluation metric the CV of the root mean square error between real and estimated densities. We show in Fig. 6 estimation results for a realistic scenario in a sizable network, consisting of a 10-km, three-lane highway stretch with several on/off-ramps, under various traffic conditions, utilizing the microscopic traffic simulation software Aimsun. It may be seen from Fig. 6 that the accuracy of the estimation scheme does not degrade substantially even for very low penetration rates, see, Fountoulakis et al. (2017) for details.

3.2. Per-lane estimation

Real-time traffic information per lane is crucial for the implementation of some novel, VACS-enabled traffic control measures, such as lane assignment and lane-change control. To the best of our knowledge, the estimation approach presented in this section is the only one in the technical literature to address the problem of per-lane estimation.

3.2.1. Traffic density dynamics

We provide the discrete-time dynamics of per-lane densities based on the conservation equation for each segment-lane cell (i, j) as

$$\rho_{i,j}(k+1) = \rho_{i,j}(k) + \frac{I}{\Delta_i} \left(q_{i-1,j}(k) - q_{i,j}(k) + L_{i,j-1 \to j}(k) + L_{i,j+1 \to j}(k) - L_{i,j \to j-1}(k) - L_{i,j \to j+1}(k) + r_{i,j}(k) - s_{i,j}(k) \right),$$
(17)

where $\rho_{i,j}$ is the total traffic density in cell (i, j); $q_{i,j}$ is the total longitudinal inflow entering cell (i + 1, j); $r_{i,j}$ and $s_{i,j}$ are the



Fig. 4. (a) Graph of system (16); (b) Graph of system (20). Green vertices denote measured states and directed arcs indicate reverse traffic flow, which exist when the corresponding elements of the matrices are nonzero.

total on- and off-ramp flows entering and exiting from cell (i, j), respectively; and $L_{i,j_1 \rightarrow j_2}$ is the total lateral (lane-changing) flow at segment *i* that enters lane j_2 from lane j_1 .

We assume in our model that the following relations hold

$$q_{i,j}(k) = v_{i,j}(k)\rho_{i,j}(k) + p_{i,j-1\to j}L_{i,j-1\to j}(k) + p_{i,j+1\to j}L_{i,j+1\to j}(k) + \bar{p}_{i,j}r_{i,j}(k),$$
(18)

$$L_{i,j_1 \to j_2}(k) = \frac{L_{i,j_1 \to j_2}^c(k)}{\rho_{i,j_1}^c(k)} \rho_{i,j_1}(k)$$
(19)

where $p_{i,j_1 \rightarrow j_2}$, $\bar{p}_{i,j} \in [0, 1]$ are parameters that indicate the percentages of "diagonal" lateral movements and ω^c represents the measurement involving only connected vehicles of variable ω . The first term in (18) is the same as in (14), while the rest terms are motivated by the fact that, at locations featuring strong lateral flows (e.g., at on-ramps or lane-drops), the flow is more accurately described by considering that a percentage of lateral or on-ramp flows acts as additional exiting longitudinal flow (i.e., flow may also move diagonally). Relation (19) is based on the reasonable assumption that the average behaviour of the population of connected vehicles in a given cell, with respect to lateral movements, is representative for the total vehicle population in that cell (for further details see, Bekiaris-Liberis et al., 2017; Papadopoulou, Roncoli, Bekiaris-Liberis, Papamichail & Papageorgiou, 2018); and the variables involved in (19) and referring to connected vehicles may be readily extracted as measurements from the connected vehicles reports.

Similarly to the case of cross-lane estimation, we can write (17)-(19) in the following form

$$\begin{aligned} x(k+1) &= A(v^{c}(k), L^{c}(k), \rho^{c}(k))x(k) + Bu(k), \\ y(k) &= C(v^{c}(k), L^{c}(k), \rho^{c}(k))x(k), \end{aligned}$$
(20)

where $x = (\rho_{1,1}, ..., \rho_{N,1}, ..., \rho_{1,M}, ..., \rho_{N,M}, r_{1,M}, ..., r_{N,M}, s_{1,M}, ..., s_{N,M})^{T}$, while *y* and *u* consist of flow measurements from fixed detectors, as in Section 3.2.3.

3.2.2. Estimation scheme

For estimation of *x* we employ a Kalman filter using model (20) and measurements v^c , L^c , ρ^c , *y*, *u*.

3.2.3. Observability and stability

As in the cross-lane case, in order to guarantee exponential stability of the Kalman filter utilized for state estimation of system (20), the most crucial fact is to show the strong structural observability of (20) Bekiaris-Liberis et al., 2017). Under



Fig. 5. Cross-lane estimation - real and estimated densities for the NGSIM case study for 5% penetration rate of connected vehicles.



Fig. 6. Cross-lane estimation – performance indices of density estimation for various penetration rates in the microscopic simulation experiments for 6 and for 12 time steps used when calculating the moving average of speed computed from connected vehicle reports.

the assumptions that the average speeds of connected vehicles are nonzero and satisfy a modified (to account for lateral flows) Courant–Friedrichs–Lewy-type condition, system ((20) (where, for simplicity, all percentage values for diagonal lateral and on-ramp flows are set to zero) is strong- structurally observable if and only if fixed flow detectors are placed at the mainstream exit and entry in each lane of the considered stretch, as well as at every cell immediately upstream of a cell with a ramp (see Fig. 4(b)). For physical systems, weak structural observability is usually sufficient for observability, which may be guaranteed when, for each pair of unmeasured ramps, an additional fixed flow sensor is placed anywhere between them (Bekiaris-Liberis et al., 2017).

3.2.4. Case studies

Real data validation using NGSIM data: We employ again NGSIM data, now dividing the stretch in four 100-m long segments. The performance of the per-lane estimation scheme is shown in Fig. 7.

Microscopic simulation-based validation: We utilize a model of the stretch of motorway A20 from Rotterdam to Gouda, in the Netherlands, (see Fig. 2(a)) implemented and validated, as outlined in Section 2.1.2, with real, lane-specific, traffic data, thus providing a realistic ground truth scenario. The estimation results are shown in Fig. 8 and indicate good estimation results even for very low penetration rates, see, Papadopoulou et al. (2018) for details.

4. Traffic flow control

VACS give rise to new opportunities for more efficient traffic control. This is due to the increased control granularity (e.g., control by lane, control by destination, control of individual vehicles) and the arbitrary space-time resolution of control measures that one may use (instead of the fixed-location traffic signs, VMS, VSL etc.). Vehicle speed control or efficient lane assignment are just two such new opportunities where automated vehicles can be exploited as actuators to improve traffic flow. Some more examples of related developments are given in what follows.

4.1. Cooperative merging

The very first article addressing the problem of efficient merging of two single-lane streams of automated vehicles was published as early as 1969 (Athans, 1969) using the then innovative



Fig. 7. Per lane estimation - real and estimated densities for the NGSIM case study for 20% penetration rate of connected vehicles.



Fig. 8. Per-lane estimation – performance indices of density estimation for various penetration rates in the microscopic simulation experiments.

LQR methodology. Some few works addressing vehicle merging appeared in the following decades, but the problem gained significance and popularity in the last decade thanks to the emergence of VACS, which led to multiple, partly overlapping approaches by various groups, see, Rios-Torres and Malikopoulos (2017) and Schmidt and Posch (2010) for dedicated informative overviews of past and more recent developments. Automated vehicle merging may be viewed as a general problem in the wider transportation domain. In the particular case of motorways, merging of vehicles entering the mainstream from on-ramps is one of the major causes of congestion. Moreover, in manual driving, the merging procedure is stressful for the drivers, since it requires a synchronized set of fast observations and actions. Therefore, the development of automated procedures for equipped vehicle merging is important, as it can increase passengers' safety and convenience and alleviate congestion.

As suggested by Ntousakis, Nikolos and Papageorgiou (2016), the overall problem complexity may be mitigated by appropriate functional decomposition. A merging location point is specified on the mainstream towards the end of the on-ramp's acceleration lane. This is the point where all merging vehicles (from either the mainstream or the on-tamp) end the merging maneuvers. All vehicles located upstream of the merging point up to a certain distance (of, e.g., some 200 m), on either the mainstream or the on-ramp, belong to the merging control region and are subject to merging control. We distinguish the following functional control layers:

- The *traffic management layer* specifies the required speed and time-headway that vehicles should have when reaching the merging point. These quantities are specified, depending on the current demand, so as to maximize safety and throughput for the merged total flow. For example, for high demand, the time-headway (which is the inverse of flow) may be set corresponding to capacity; while for lower demand, there is a range of possible settings to select from according to other considerations.
- The merging sequence layer specifies the sequence of vehicles in the control region in reaching the merging point. There are

various ways of doing this, such as: FIFO; according to current vehicle speeds and distances; optimal. The MS is updated regularly, e.g., whenever a new vehicle enters the control region. Once a MS is specified, each vehicle obtains a putative leader, which is the vehicle to merge ahead of it. A putative leader may or may not be the physical leader (i.e., the same-lane vehicle ahead).

- The vehicle control layer calculates a longitudinal vehicle trajectory up to the merging point by solving a finite-horizon optimal control problem. Specifically, based on a simple model of vehicle kinematics, the trajectory problem consists in moving the vehicle from its current state to the final state, at which: final position is the merging point, final speed is the one prescribed by traffic management, and final acceleration and jerk are zero. The time horizon is fixed according to the MS, the specified time-headway and the merging time of the putative leader. A quadratic criterion leads to smooth, passenger-convenient and fuel-saving trajectories. Specifically, the minimization of acceleration is directly connected to the minimization of fuel consumption, while the minimization of jerk is connected to passenger comfort. The overall problem is readily solved analytically or via a time-variant LQR.
- The vehicle trajectory problem must be solved repeatedly (MPC approach), using updated measurements in real time, until the merging procedure is finalized. Application of MPC is necessary due to many potential disturbances or changes, such as: physical leader constraints, putative leader trajectory changes, insufficient actual trajectory tracking, MS update, final speed and headway update.

As it was demonstrated by Ntousakis, Porfyri, Nikolos and Papageorgiou (2014), even relatively simple MS schemes may produce smooth merging and efficient total merged traffic flow, although partial differences in the resulting behaviour are also observed. More specifically, Ntousakis et al. (2014) tested and compared two MS algorithms. In a first algorithm, vehicles are merging in the same sequence as they are entering the control region (FIFO). In a second algorithm, vehicles are merging in a sequence which depends on the time they need to arrive to the merging point, assuming constant (current) speed inside the control region. Both algorithms specify and update (at each time step) the MS. It was assumed that all vehicles in the network are equipped with ACC systems and are enabled with V2V communication capabilities. The microscopic simulator Aimsun was used to perform the corresponding simulations for testing and demonstration. Both algorithms provided similar results concerning the macroscopic values of total merged flow and density. However, the second algorithm is closer to real driving behaviour, since it evaluates also forward gaps and mitigates unnecessary decelerations.

4.2. Vehicle trajectory control

Fully automated vehicles could improve safety and efficiency of traffic operations by reducing accidents caused by human driver errors, improve driver and passenger comfort and, at the same time, help to reduce traffic congestion. The development of fully automated driving systems is inherently related to planning and updating a vehicle path, which should be safe, collision-free, user-acceptable, as well as efficient for the vehicle itself, but also for the emerging traffic flow. Planning such a path can be seen as a trajectory generation problem, i.e., creation of a quasi-continuous sequence of states that must be tracked by the vehicle control.

The complexity of real-world environments, avoidance of static and moving obstacles, compliance with traffic rules and consideration of human driving behaviour aspects are some of the factors that make automated vehicle trajectory control a challenging problem. More in particular, the trajectory generation process must not only be effective according to a variety of criteria, but also perform real-time adaptations to an ever-changing environment, with surrounding vehicles continuously deviating from their projected trajectories. Last not least, as the penetration rate of automated vehicles increases, their behaviour should account not only for individual criteria (selfish behaviour), but also for the efficiency of the emerging traffic flow.

Potential field methods generate a two-dimensional (road surface) field, based on the design of appropriate potential functions for obstacles, road structures, traffic regulations and the goals to be achieved. Then, a vehicle trajectory is generated by moving, at any road location towards the maximum descent direction (gradient) of the field (Wang, Wu & Li, 2015; Wolf & Burdick, 2008). Ji, Khajepour, Melek and Huang (2017) proposed a trajectory generation method along with a model-predictive path tracking controller. However, the potential field methods do not behave well in the presence of mobile obstacles, as they do not consider future path information regarding the own and other vehicles movements (Shum, Morris & Khajepour, 2015). Due to the fact that the system dynamics and the movement of other vehicles are not considered during the trajectory generation, the planned path may in some situations turn out to be non-feasible to be tracked by the vehicle (Erlien, 2015). Potential field methods are quite popular due to the intuitive problem formulation and their low computational cost (Rasekhipour, Khajepour, Chen & Litkouhi, 2017), even with complex potential functions. However, there are still open issues regarding the integration of system dynamics and constraints within such methods that were discussed some three decades ago (Koren & Borenstein, 1991).

This drawback can be alleviated by formulating the trajectory generation problem as an optimal control problem. Such approaches allow for concurrent consideration of system dynamics and anticipated obstacle movements during the trajectory generation process, thus avoiding myopic actions. For example, in Carvalho, Gao, Lefevre and Borrelli (2014), the distance between the vehicle and an obstacle is used to generate approximate linear constraints for obstacle avoidance, while in Gao, Gray, Tseng and Borrelli (2014), the obstacles are considered as ellipse-shaped constraints, in order to keep the automated vehicle robustly far from them. In the same spirit, Gao, Lin, Borrelli, Tseng and Hrovat (2010) included in the cost function factors dependent on the longitudinal distance between the automated vehicle and the obstacles, in order to generate a collision-free path. In some works, the merits of both potential field and optimal control path planning techniques are combined. Rasekhipour et al. (2017) and Wang et al. (2019) developed an MPC controller for path planning of autonomous vehicles, which avoids obstacles and observes road regulations by introducing appropriate penalty functions in the objective function of the optimal control problem.

Optimal control approaches usually map the optimal control problem to a NLP problem that can be solved using numerical NLP solvers, see for example, Gray, Ali, Gao, Hedrick & Borrelli (2012), Werling and Liccardo (2012) and Ziegler, Bender, Dang and Stiller (2014). The main drawback of this mapping is that it produces a locally optimal solution, the quality of which is very much dependent on the initial guess trajectory employed.

Another attractive alternative for solving optimal control problems is the use of DP techniques (Bertsekas, 2005). DP can be applied to a broad range of system models and allows for easy integration of system and obstacle constraints. In addition, DP techniques produce a globally optimal solution for the optimal control problem. Unfortunately, due to the curse of dimensionality (Bellman, 1954), DP techniques are restricted to small-scale problems. For this reason, they are sometimes combined with



Fig. 9. In a simple highway stretch and for various traffic loads, vehicles following our path-planning approach yield a lower deviation from optimal travel time (with everyone moving at their desired speed) than conventional vehicles.

other optimization methods. For example, Tanzmeister, Friedl, Wollherr and Buss (2013) used DP to generate a rough reference vehicle trajectory for parking scenarios, where all obstacles are considered static.

Makantasis and Papageorgiou (2018) derived an algorithm for generating a feasible path for an automated vehicle from the opportune formulation of a discrete-time optimal control problem. Vehicle dynamics form the state equations of the problem; while the road geometry, the presence of obstacles (i.e., other moving vehicles and road-boundaries) and traffic rules are taken into account via appropriate potential-field like functions to ensure path feasibility. Exploiting the structure of the system dynamics (state equations), a reduced gradient may be readily obtained, and, thereby, the optimal control problem is mapped to a NLP problem in the reduced space of the control variables. Starting from an arbitrary initial solution (or path), a local minimum is obtained with a very efficient iterative FDA (Papageorgiou, Marinaki, Typaldos & Makantasis, 2016). A simplified DP algorithm is also conceived to deliver the initial solution, greatly enhancing the quality of obtained local minima.

Based on the above work, Typaldos, Mountakis, Papageorgiou and Papamichail (2019) take advantage of low computation times to embed this optimization-based path-planning approach within a MPC framework, which is implemented in Aimsun's micro-simulation platform (TSS-Transport Simulation Systems, 2014). Considering a homogeneous motorway stretch and alongside other vehicles following Aimsun's default driving behaviour, one or more vehicles are instructed to track a path produced by the MPC-based optimization approach. The path for each such controlled vehicle is generated according to the current lane and speed of surrounding vehicles and is re-generated online in case of substantial changes. Aimsun's micro-simulation platform enables a thorough experimental evaluation of the approach, by creating a huge number of different traffic scenarios as the vehicles drive in the simulated road environment. In addition, this experimental framework allows the investigation of the impact of the suggested approach on the traffic flow as a whole for increasing penetration rates of automated vehicles. Some first results (see Fig. 9) indicate faster advancement of the ego vehicle compared to others, particularly in free-flow conditions.

Ongoing work delivers further improvements to the open-loop path-planning method by employing state-of-the-art non-convex optimization techniques, such as sequential convex programming (Lipp & Boyd, 2016), for the solution of the optimization problem.



Fig. 10. Illustration of the control strategy operation.

In future work, we aim for a thorough investigation of the effects of fully automated vehicles (guided in a decentralized way by our approach) on traffic flow as a function of penetration rate. In addition, it is interesting to design a common optimal control problem for the simultaneous path-planning of multiple vehicles to study the synergistic effects on overall traffic flow. Finally, we aim to design optimization objectives or weights that will not merely serve each individual automated vehicle, but also contribute to a more efficient traffic flow (e.g., focusing on mitigating congestion).

4.3. Real-time ACC time-gap control

The ACC systems have been designed to increase the driving safety and comfort. However, if conservative values are set for their parameters (e.g., the time-gap), then the ACC systems may actually lead to a deterioration rather than improvement of traffic flow efficiency, as already seen in Fig. 1. By the same token, lower than manual time-gap settings would lead to increased motorway capacity per lane and decreased capacity drop at the head of congestion, and this impact can be exploited for increased traffic flow efficiency if the settings of the ACC-vehicles could be updated dynamically in real time through the operation of an ACC-based control strategy, see, Kesting, Treiber, Schönhof and Helbing (2008).

For the design of such a control strategy (Spiliopoulou, Manolis, Vandorou & Papageorgiou, 2018), consider a motorway with both manually-driven and ACC-vehicles. The ACC-vehicle drivers may introduce their desired ACC system settings, i.e., desired speed, v_d , and minimum time-gap, T_d , but these settings are subject to change if the control strategy recommends (or orders) different values. The motorway is considered to be divided in sections, and the traffic management centre applies the proposed control strategy at every motorway section i = 1, 2, ... independently, as illustrated in Fig. 10, in dependence of the current traffic conditions. In particular, at every control interval t_c , the strategy receives real-time measurements (or estimates) of the exiting flow q_i and mean speed v_i of every section *i*to decide on appropriate settings. The control strategy determines in real time the time-gap of the ACC vehicles that lead to increase of the static and dynamic road capacity, but this is indeed only done where and when needed. The proposed strategy has two goals which are presented below.

Capacity increase: To increase the (static) capacity, the strategy gradually decreases the suggested time- gap, when traffic flow is approaching the nominal capacity. In particular, the strategy



Fig. 11. Calculation of the suggested time-gap value for real-time capacity increase.

calculates the suggested time-gap as a monotonically decreasing function of the current section flow, $T[q_i(k)]$, as shown in Fig. 11. Note that the suggested time-gap value is reduced to the minimum value before the flow reaches the nominal capacity of the section. In this way, the strategy aims to delay, or even prevent, the formation of congestion, by maximizing timely the section's capacity. If, despite the capacity increase, the section becomes congested (e.g., due to even higher arriving demand or due to a shockwave arriving from downstream), then the strategy activates its second goal.

Discharge flow increase: The second goal of the proposed control strategy is the maximization of the discharge flow during congestion (dynamic capacity). It is empirically known that the discharge flow at the head of a congested area is lower than capacity, and the second goal is to mitigate this capacity drop. The strategy first identifies the location of active bottlenecks (congestion head), then suggests the minimum admissible time-gap T_{min} for all ACC vehicles in the sections just upstream and just downstream of the congestion head. In that way, the strategy achieves the increase of the discharge flow at the congestion head.

The above control decisions are summarized in the following equation which determines the suggested time-gap,

$$T_{stg,i}(k) = \begin{cases} T_i[q_i(k)] & \text{if } v_i(k) > v_{cong} \\ T_{min} & \text{at active bottlenecks} \\ T_{max} & \text{else} \end{cases}$$
(21)

where k = 1, 2, ..., is the discrete time index, and v_{cong} indicates the congestion limit. The control strategy decisions are calculated externally, at an infrastructure-based traffic management centre (or road-side unit) and are disseminated to the ACC-vehicles, e.g., via V2I communication. The ACC-vehicles receive the suggested time-gap, but they apply it only if their individual time-gap setting, T_d , is higher than the time-gap calculated by the controller.

The proposed strategy was tested through microscopic simulation in Aimsun using the real motorway stretch of Section 2.1.2, where congestion is created due to an on-ramp bottleneck. The simulation results showed that, the higher the penetration rate, the bigger the improvement of the traffic conditions. Figs. 12 and 13 present the spatio-temporal diagrams of speed, considering various penetration rates, for the no-control case and the strategy application case, respectively. It can be seen that, for the no-control case, the conservative ACC time-gap values lead to an increasingly strong congestion as the penetration rate of ACC-vehicles increases. On the contrary, in the control case, the control strategy achieves significant improvement even for low penetration rates (e.g., 5%), while for high penetration rates the congestion is almost resolved.

4.4. Lane-change control

4.4.1. Introduction

Lane-change control is a promising feature that can be exploited for traffic management at merging bottlenecks (e.g.,

lane-drops, on-ramp merges), where human drivers usually perform suboptimal lane-changes based on erroneous perceptions, which may trigger premature congestion; but even at non-merge bottlenecks (e.g., bridge, tunnel, strong curvature, grade), congestion usually starts at one lane, while other lanes still have capacity reserves, hence better lane distribution of vehicles may increase the cross-lane capacity. In the presence of a sufficient percentage of vehicles equipped with lane-changing automatic controllers or advisory systems, the overall throughput at the bottleneck location may be improved by execution of specific lane-changing commands dictated by a central decision maker.

Early works addressed the lane assignment problem for fully automated highway systems. The seminal work by Varaiya (1993) proposed a hierarchical framework for a fully automated motorway, in which the decisions on the lane-changing behaviour of vehicles are addressed in the link layer. Following this framework, several strategies have been proposed to solve the problem of lane assignment in the link layer; strategies include designing control methods suitable for real-time applications, with the development of structured heuristic rules (Rao & Varaiya, 1994); implementing lane-routing algorithms (Lee & Lee, 1997); and defining control laws to stabilize traffic conditions (Li, Horowitz, Alvarez, Frankel & Robertson, 1997). Optimization methods for path planning through lanes have also been developed, but the computation complexity of the proposed optimization problems makes them hardly applicable in a real-time context (Hall & Caliskan, 1999; Hall & Lotspeich, 1996; Kim, Cho & Medanić, 2005; Ramaswamy, Medanic, Perkins & Benekohal, 1997).

More recently, in a mixed traffic context with conventional and VACS-equipped vehicles, lane-changing control has been considered, together with variable speed limits and ramp metering, in integrated traffic management strategies (see Section 4.5). Also, a combined lane-changing and VSL control strategy was developed by Zhang and Ioannou (2017) with the purpose of avoiding lane changes in the immediate proximity of a bottleneck, which, especially in the case of heavy vehicles, may lead to premature triggering of congestion. In particular, lane-changing commands delivered as recommendations to drivers are delivered in real time according to a set of case-specific rules. Schakel and van Arem (2014) proposed a system that aims at improving traffic conditions by sending advice on lane, speed, and headway to vehicles equipped with an in-car advisory system. The advice is computed at a traffic management centre based on a lane-level traffic state prediction model. Furthermore, Guériau, Billot, Faouzi, Hassas and Armetta (2015) proposed a multi-agent decentralized framework with the aim of performing cooperative lane-changing tasks according to information exchange between vehicles and a roadside unit located in the proximity of a bottleneck.

An optimal feedback control strategy formulated as a LQR was recently proposed by Roncoli, Bekiaris-Liberis and Papageorgiou (2016a) and Roncoli, Bekiaris-Liberis and Papageorgiou (2017). The resulted linear state-feedback control law is highly efficient in real time even for large-scale networks. In contrast to other approaches, this strategy is based on a rigorous application of LQR methodology and does not involve heuristic rules. The control strategy aims at regulating the lane assignment of vehicles upstream of a bottleneck location so as to maximize the bottleneck throughput, targeting critical densities as set points.

4.4.2. Linear multi-lane traffic flow model

We consider a multi-lane motorway subdivided into segments of length L_i , where each segment is composed of $M_i - m_i + 1$ lanes (m_i and M_i being the minimum and maximum indices of lanes in segment *i*) indexed by *j*, denoting each element of the resulting grid as "cell", indexed by (*i*, *j*). The model is formulated considering the discrete time step *T*, indexed by k = 0, 1, ... We



Fig. 12. No-control case: spatio-temporal diagrams of speed considering various PRs.



Fig. 13. Traffic control case: spatio-temporal diagrams of speed considering various PRs.

define traffic density $\rho_{i, j}$, whose dynamics evolves according to Roncoli et al. (2016a):

$$\rho_{i,j}(k+1) = \rho_{i,j}(k) + \frac{T}{L_i} \left(q_{i-1,j}(k) - q_{i,j}(k) + f_{i,j-1}(k) - f_{i,j}(k) + d_{i,j}(k) \right)$$
(22)

where $q_{i,j}$ is the longitudinal flow from cell (i, j) to cell (i + 1, j); $f_{i,j}$ is the net lateral flow from cell (i, j) to cell (i, j + 1) (i.e., from right to left lanes); and $d_{i,j}$ is the external flow entering the network in cell (i, j). Considering the well-known relation $q_{i,j} = \rho_{i,j}v_{i,j}$ and assuming that speeds $v_{i,j}$ are known parameters,

$$x(k+1) = A(k)x(k) + Bu(k) + d(k),$$
(23)

where
$$x = [\rho_{0,m_0} \dots \rho_{N,M_N}]^I$$
, $u = [f_{0,m_0} \dots f_{N,M_N-1}]^I$, and $d = [\frac{T}{L_0} d_{0,m_0} \dots \frac{T}{L_N} d_{N,M_N}]^T$.

4.4.3. LQR problem formulation

The proposed control actions are intended for usage prior to the possible onset of congestion, aiming to delay or avoid it. Therefore, we may assume that the overall traffic flow entering the controlled area does not yet exceed the bottleneck capacity, and, hence, that the speed in all cells is high and roughly constant, which implies that (23) can be written as an LTI system

$$x(k+1) = Ax(k) + Bu(k) + d(k).$$
 (24)

We define the following quadratic cost function, over an infinite time horizon, that accounts for the penalization of the difference between some selected densities and pre-specified (constant) set-point values, as well as a penalty term aiming at maintaining small control inputs, i.e., small lateral flows:

$$J = \sum_{k=0}^{\infty} \left\{ \left[Cx(k) - \hat{y} \right]^T Q \left[Cx(k) - \hat{y} \right] + u^T(k) R u(k) \right\},$$
 (25)

where $Q = Q^T \ge 0$ and $R = \varphi I > 0$ are weighting matrices associated to the state tracking error and control actions, respectively, while *C* reflects the cells that are tracked, namely the cells at the bottleneck location for which set-points \hat{y} are used.

4.4.4. Stabilizability, detectability, and regulator

The problem of minimizing (25) subject to (24) can be solved through an LQR, which provides a stabilizing feedback control law if the original system is, at least, stabilizable and detectable (Lewis, Vrabie & Syrmos, 2012). Since eigenvalues λ of matrix *A* are equal to the elements in the main diagonal (*A* is, by construction, lower triangular) and *v* is always positive, the only marginally stable modes are the ones related to segments without any other segment downstream (i.e., at a lane-drop). Applying the Hautus-test (Lewis et al., 2012) to marginally stable modes for the pair (*A*, *B*), we see that stabilizability is trivially satisfied for any system following the defined network structure. On the other hand, testing the pair (*A*, *C*) for detectability results in the requirement for an arbitrary set-point also for cells that do not have any other cell downstream.

The solution to the proposed LQR problem is the linear feedback/feedforward control law

$$u^{*} = -Kx + u_{\rm ff},$$

$$u_{\rm ff} = \left(R + B^{T} P B\right)^{-1} B^{T} \left(I - (A - BK)^{T}\right)^{-1} \left(C^{T} O \hat{v} - P d\right).$$
(26)
(27)

$$u_{\rm ff} = (R + B^T P B)^{-1} B^T (I - (A - BK)^T)^{-1} (C^T Q \hat{y} - P d), \qquad (27)$$

where the feedback gain matrix *K* is defined according to classic LQR theory and can be computed offline by using the solution *P* of the Algebraic Riccati Equation Williams & Lawrence, 2007). Note that in the LQR derivation we assumed that the external flows *d* are constant, but ((27), which is an affine function of *d*, may be computed online for (slowly) time-varying inflows.

4.4.5. Optimal set-point tuning via extremum seeking

The set-points \hat{y} should typically be set equal to the critical density values (for each lane), at which the flows are maximum. If these values are not known reliably at the bottleneck location, one may employ discrete-time extremum seeking (Anderson & Moore, 1971) for tuning set-points, aiming at achieving an optimal value of a cost (flow maximization) utilizing only real-time measurements. We use as a cost function TTT over a finite time horizon K, as defined by Ariyur and Krstić (2003). Experimental results show a fast convergence to the optimum TTT, corresponding to set-points equal to critical densities, as can be seen in Fig. 14.

4.4.6. Policy-based density distribution at bottlenecks

As a result of implementing the control law 26) and (27), the traffic density distribution among different lanes remains (roughly) constant under any demand scenario. Although this behaviour would not produce any negative impact on the traffic performance, it may be, in some circumstances, undesirable to road traffic authorities, who may, for various reasons (e.g., because of lane usage regulations), prefer different specific lane distributions for different flow demand. In such cases, the control law



Fig. 14. Cost function value achieved by applying the extremum seeking algorithm.

may be switched on only when the flow approaches capacity and congestion seems imminent. Alternatively, one may extend the control strategy so that, besides aiming at tracking the critical density (e.g., when demand is close to bottleneck capacity), it also aims at distributing the vehicles in the bottleneck area among the different lanes according to a given policy. This is achieved by modifying the feed-forward term of control law ((27) as follows

$$u_{\rm ff}(k) = \left(R + B^T P B\right)^{-1} B^T (I - (A - BK)^T)^{-1} \left(C^T Q \ \psi(d(k)) - P \ d(k)\right),$$
(28)

where the function ψ defines the pursued lane distribution policy.

The reported results (Fig. 15 and Fig. 16) were obtained by use of macroscopic simulation, whereby the controller decisions are directly applied to the simulator as macroscopic lateral flows. For practical applicability of the method, the macroscopic lateral flows need to be converted to individual messages to be submitted to a corresponding number of connected vehicles, something that does not create any major methodological difficulties. In fact, such a realization was developed by Markantonakis, Skoufoulas, Papamichail and Papageorgiou (2019), where the lane-change control method was combined with mainstream flow control (implemented by use of vehicle speed control) and was tested in microscopic simulation.

4.5. Integrated motorway network control

As discussed, the use of VACS permits the implementation of an increased range of control tasks, including vehicle-actuated variable speed limits and lane-changing control, which produce unprecedented capabilities for link-level motorway control. We present here a model-based control approach, which addresses simultaneously, in an integrated way, multiple control measures for a whole motorway link or network through the formulation of a linearly-constrained convex QP problem, based on a firstorder multi-lane model that has been appropriately designed (see Section 2.2.2).

4.5.1. Optimal control problem formulation

We subdivide a given motorway stretch into segments i = 1, ..., I and lanes j = 1, ..., J, while considering a discrete time index k = 1, ..., K, where *K* is the optimization horizon. We manipulate the following three control actions, where each action ω has a different index k^{ω} reflecting different control steps:

- RM, described by $r_{i, j}(k^R)$, that represents the inflow from the on-ramps to the motorway mainstream.
- MTFC, described by $q_{i,j}(k^Q)$, that represents the mainstream flow from a cell to the downstream one, which could be implemented via variable speed limits.



Fig. 15. Density contour plots for: (a) the no-control case; and (b) the control case applying strategy 26) and ((27) with constant set-points.



Fig. 16. The flow exiting the bottleneck area for the feedback control strategy: (a) with constant set-points; and (b) for policy-based density distribution.

- LCC to define ordered lateral flows for each segment-lane, enabling an optimal distribution of traffic flow among the different lanes; the corresponding control variables are $f_{i,j,\bar{j}}(k^F)$, denoting the flow of vehicles moving from lane j to lane \bar{j} $(\bar{j} = j \pm 1)$ of segment i.

The dynamic equation for densities $\rho_{i,j}$ is given by the following conservation equation:

$$\rho_{i,j}(k+1) = \rho_{i,j}(k) + \frac{T}{L_i} \bigg[q_{i-1,j}(k^{\mathbb{Q}}) + r_{i,j}(k^{\mathbb{R}}) - q_{i,j}(k^{\mathbb{Q}}) - \gamma_{i,j}(k) \sum_{j=1}^{J} q_{i-1,j}(k^{\mathbb{Q}}) + f_{i,j+1,j}(k^{\mathbb{F}}) + f_{i,j-1,j}(k^{\mathbb{F}}) - f_{i,j,j-1}(k^{\mathbb{F}}) - f_{i,j,j+1}(k^{\mathbb{F}}) \bigg]$$
(29)

where $\gamma_{i,j}(k)$ are estimated turning rates at off-ramps. RM actions may lead to the creation of queues at on-ramps $w_{i,j}(k)$, modelled by the following dynamics, where $d_{i,j}(k)$ is the external demand

feeding the ramp queue

$$w_{i,j}(k+1) = w_{i,j}(k) + T \Big[d_{i,j}(k) - r_{i,j}(k^R) \Big].$$
(30)

Longitudinal flows are modelled via a piecewise-linear FD, with an additional linear function to reflect the capacity drop phenomenon if the upstream density is over-critical (Roncoli et al., 2015b), although alternative approaches may be employed, as shown in Kontorinaki et al. (2017), see Section 2.2.2. Since longitudinal flows are assumed controllable via corresponding lowering of equipped vehicle speeds, the FD is converted to upper bounds for the controllable longitudinal flows as follows.

$$q_{i,j}(k) \le \nu_{i,j}^{free} \rho_{i,j}(k), \tag{31}$$

$$q_{i,j}(k) \le -\frac{\nu_{i,j}^{free}\rho_{i,j}^{cr} - q_{i,j}^{jam}}{\rho_{i,j}^{jam} - \rho_{i,j}^{cr}}\rho_{i,j}(k) + \frac{\rho_{i,j}^{cr}(\nu_{i,j}^{free}\rho_{i,j}^{jam} - q_{i,j}^{jam})}{\rho_{i,j}^{jam} - \rho_{i,j}^{cr}}, \quad (32)$$

$$q_{i,j}(k) \le v_{i+1,j}^{free} \rho_{i+1,j}^{cr}, \tag{33}$$



Fig. 17. Speed contour plots: (a) for the no-control case; and (b) the optimized case; of case study 1 for on-ramp demand increased by 30%.

$$q_{i,j}(k) \le -\frac{\nu_{i+1,j}^{free}\rho_{i+1,j}^{cr}}{\rho_{i+1,j}^{jam}-\rho_{i+1,j}^{cr}}\rho_{i+1,j}(k) + \frac{\nu_{i+1,j}^{free}\rho_{i+1,j}^{cr}\rho_{i+1,j}^{jam}}{\rho_{i+1,j}^{jam}-\rho_{i+1,j}^{cr}},$$
(34)

where $v_{i,j}^{free}$ is the free-flow speed, $\rho_{i,j}^{cr}$ is the critical density, $\rho_{i,j}^{jam}$ is the maximum admissible density, and $q_{i,j}^{jam}$ is the minimum flow achievable due to the capacity drop.

The computation of lateral flows is fully delegated to the optimizer, while only upper bounds are specified to the non-negative lateral flows as follows:

$$\left[f_{i,j,j-1}(k^{F}) + f_{i,j,j+1}(k^{F})\right] \le \frac{L_{i}}{T}\rho_{i,j}(k),$$
(35)

$$\left[f_{i,j-1,j}(k^{F}) + f_{i,j+1,j}(k^{F})\right] \le \frac{L_{i}}{T} \left[\rho_{i,j}^{jam} - \rho_{i,j}(k)\right],$$
(36)

$$f_{i,j,j-1}(k^F) \le f^{\max},\tag{37}$$

$$f_{i,j,j+1}(k^F) \le f^{\max}.$$
(38)

Finally, we employ a quadratic cost function that includes (Roncoli, Papageorgiou & Papamichail, 2015c).

- Linear terms reflecting TTS, as the most crucial control objective, which includes both the total travel time on the mainstream and the total waiting time at on-ramp queues, and a penalty term to avoid excessive lateral (lane-changing) flows at specific locations.
- Quadratic penalty terms to reduce time variations of RM and LCC control variables, as well as to reduce time and space fluctuations of the speed values (approximated via appropriate linearized expressions).
- The resulting optimization problem has a convex QP form, which allows achieving a global optimum (in contrast to other nonlinear approaches) in very low computation time also for large-scale systems.

4.5.2. Case study 1: open-loop optimization

We implement our optimization problem for a 4-lane network stretch of 5.26 km in length, which includes three on-ramps and three off-ramps, of the Monash Freeway (M1) around Melbourne, Australia. We investigate the morning peak period (5 a.m.–9 a.m.), where a major congestion appears due to high demand and weaving close to the ramps. First, the model parameters of the multi-lane traffic flow model were calibrated by use of real traffic

data (Roncoli et al., 2015b) (no-control case); then, the optimization problem, with tuned parameters was applied for the entire 4-hour horizon. We observe that the TTS improves by 23% with respect to the no-control case, employing only RM and LCC actions. In addition, we test a scenario where the on-ramp demands are increased by 30% and the resulting optimization shows a 52% improvement of TTS with respect to the corresponding no-control case (see Fig. 17), with the use of all three control actions; see, Roncoli, Papageorgiou and Papamichail (2015a) for more details.

4.5.3. Case study 2: model predictive control in microscopic simulation

We perform further experiments in an MPC framework using the microscopic traffic simulator Aimsun, with opportunely modified car-following and lane-changing models. We implemented the model of a stretch of motorway A20 from Rotterdam to Gouda in the Netherlands, validated with real traffic data (as discussed in Section 2.1.2) featuring a daily congestion pattern (see Fig. 18, top). The stretch contains all ingredients of a complex infrastructure (lane-drop, on- and off-ramps). We implement appropriate algorithms to translate optimal flows to speed or lane-changing commands. Results show the capability of the proposed method to improve traffic conditions, via flexible coordination of all available control actions (see Fig. 18, middle), as well as when excluding RM from the set of control actions (see Fig. 18, bottom). In addition, in Roncoli, Papamichail and Papageorgiou (2015), we evaluate a scenario where different percentages of vehicles are equipped with ACC systems, showing that the detrimental effects due to the choice of a high headway in ACC systems (capacity reduction) may be mitigated by using an appropriate control strategy acting at a macroscopic level.

4.6. Other developments

Further interesting developments of vehicle-based traffic control, beyond those addressed in previous sections, concern the design of eco-driving vehicle trajectories (Typaldos, Papamichail & Papageorgiou, 2018) and of string-stable ACC control laws (Bekiaris-Liberis, Roncoli & Papageorgiou, 2018).

Another stream of work extends and applies recent advanced nonlinear feedback concepts to a variety of motorway traffic control problems. A number of related significant and rigorous results are of significance for both conventional and VACS-



Fig. 18. Contour plots of the speed in the no-control case (top), applying MPC considering all control actions for 50% penetration rate (middle), and applying MPC without RM (bottom) in case study 2.

including traffic flow (Karafyllis, Kontorinaki & Papageorgiou, 2017; Kontorinaki, Karafyllis & Papageorgiou, 2019). This work stream may indeed attract more researchers to adopt the proposed paradigm for nonlinear feedback traffic control. Finally, motorway traffic control with novel PDE-based approaches delivered preliminary, but promising results (Karafyllis & Papageorgiou, 2019; Karafyllis, Bekiaris-Liberis & Papageorgiou, 2018).

5. Conclusions

The emergence of novel technologies pertaining to vehicle automation and connectivity at various levels bears some risks with regard to their impact on the emerging traffic flow efficiency; but also offers unprecedented opportunities for safer and more efficient traffic flow. The latter is conditioned by the development of appropriate traffic management concepts that exploit the available and forthcoming novel features in a multitude of ways. This paper presented an overview of developments and needs regarding motorway traffic management in presence of VACS, with particular focus on achievements obtained within TRAMAN21.

TRAMAN21 was the first ERC Advanced project to address traffic management and produced a number of innovative concepts, tools and results that open up new horizons for traffic management research and practice in presence of VACS. In a first step, one needs to gain awareness about the opportunities offered by novel vehicle technologies, hence an extensive review established the set of VACS features, which are most relevant from a motorway traffic management perspective. Traffic flow modelling in presence of VACS is a prerequisite for the design and testing of efficient traffic management approaches and must be addressed at two levels, microscopic and macroscopic, for a variety of utilizations. A microscopic simulator in presence of VACS was outlined as an add-on of a commercial conventional simulator by appropriately modifying the vehicle behaviour to reflect the impact of selected VACS. At the macroscopic level, novel macroscopic traffic flow models were presented, able to capture multi-lane highway traffic as well as traffic comprising VACS-equipped vehicles at arbitrary penetration levels.

In the evolving traffic environment with VACS, equipped vehicles may act as moving sensors as well as actuators, exhibiting a driving behaviour that improves the emerging traffic flow and executing orders and advice received from the traffic control centre so as to maximize network efficiency and minimize congestion. To this end, innovative traffic estimation and control methods and tools were presented and tested in simulation. In particular, traffic control was addressed at vehicle level, local level and network level and comprises unconventional actions that are not feasible by use of the current road-side actuator technology.

It is worth mentioning at this point that, in the frame of the EU H2020 project INFRAMIX (www.inframix.eu), the presented estimation tools and some of the control methods described in this paper are being tested in real traffic conditions.

Funding

This work was supported by the European Research Council under the European Union Seventh Framework Programme (FP/2007-2013)/ERC Advanced Grant Agreement n. 321132, project TRAMAN21. Nikolaos Bekiaris-Liberis was also supported by the European Commission's Horizon 2020 research and innovation program under the Marie Sklodowska-Curie Grant Agreement n. 747898, project PADECOT.

References

Anderson, B.D.O., & Moore, J. B. (1971). Linear Optimal Control.

- Aria, E., Olstam, J., & Schwietering, C. (2016). Investigation of automated vehicle effects on driver's behavior and traffic performance. In *Transportation research* procedia: 15 (pp. 761–770). Elsevier B.V. doi:10.1016/j.trpro.2016.06.063.
- Ariyur, K. B., & Krstić, M. (2003). Real-time optimization by extremum-seeking control. Hoboken, NJ: John Wiley & Sons, Inc.
- Arnaout, G., & Bowling, S. (2011). Towards reducing traffic congestion using cooperative adaptive cruise control on a freeway with a ramp. *Journal of Industrial Engineering and Management*, 4(4), 699–717. doi:10.3926/jiem.344.

Athans, M. (1969). A unified approach to the vehicle-merging problem. *Transportation Research*, 3(1), 123–133. doi:10.1016/0041-1647(69)90109-9.

- Bayar, B., Sajadi-Alamdari, S. A., Viti, F., & Voos, H. (2016). Impact of different spacing policies for adaptive cruise control on traffic and energy consumption of electric vehicles. In Proceedings of the 24th Mediterranean conference on control and automation, MED (pp. 1349–1354). Institute of Electrical and Electronics Engineers Inc. doi:10.1109/MED.2016.7535939.
- Bekiaris-Liberis, N., Roncoli, C., & Papageorgiou, M. (2016). Highway traffic state estimation with mixed connected and conventional vehicles. *IEEE Transactions* on Intelligent Transportation Systems, 17(12), 3484–3497. doi:10.1109/TITS.2016. 2552639.
- Bekiaris-Liberis, N., Roncoli, C., & Papageorgiou, M. (2017). Highway traffic state estimation per lane in the presence of connected vehicles. *Transportation Research Part B Methodological*, 106, 1–28, doi:10.1016/LTRB.2017.11.001
- Part B Methodological, 106, 1–28. doi:10.1016/J.TRB.2017.11.001.
 Bekiaris-Liberis, N., Roncoli, C., & Papageorgiou, M. (2018). Predictor-based adaptive cruise control design. *IEEE Transactions on Intelligent Transportation Systems*, 19(10), 3181–3195. doi:10.1109/TITS.2017.2771501.
- Bellman, R. (1954). The theory of dynamic programming. Bulletin of the American Mathematical Society, 60(6), 503–516. doi:10.1090/s0002-9904-1954-09848-8.
 Bertsekas, D. P. (2005). Dynamic programming and optimal control: I (3rd ed.). Bel-
- mont, Massachusetts: Athena Scientific.
- Carvalho, A., Gao, Y., Lefevre, S., & Borrelli, F. (2014). Stochastic predictive control of autonomous vehicles in uncertain environments. In Proceedings of the international symposium on advanced vehicle control (AVEC) Retrieved from http://www.researchgate.net/profile/Stephanie_Lefevre3/ publication/263308694_Stochastic_predictive_control_of_autonomous_vehicles_ in_uncertain_environments/links/0f31753a85c205a92c000000.pdf.
- Cha, S. H., Rotkowitz, M., & Anderson, B. D. O. (2008). Gain scheduling using timevarying Kalman filter for a class of LPV systems. In *Proceedings of the 17th IFAC* world congress. doi:10.3182/20080706-5-KR-1001.2157.
- Daganzo, C. F. (1994). The cell transmission model: A dynamic representation of highway traffic consistent with the hydrodynamic theory. *Transportation Research Part B*, 28(4), 269–287. doi:10.1016/0191-2615(94)90002-7.
- Delis, A. I., Nikolos, I. K., & Papageorgiou, M. (2014). High-resolution numerical relaxation approximations to second-order macroscopic traffic flow models. *Transportation Research Part C Emerging Technologies*, 44, 318–349. doi:10.1016/j.trc. 2014.04.004.
- Delis, A. I., Nikolos, I. K., & Papageorgiou, M. (2015a). Macroscopic modelling and simulation of multi-lane traffic. In Proceedings of the IEEE 18th international conference on intelligent transportation systems (pp. 2213–2218). IEEE. doi:10.1109/ ITSC.2015.357.
- Delis, A. I., Nikolos, I. K., & Papageorgiou, M. (2015b). Macroscopic traffic flow modeling with adaptive cruise control: Development and numerical solution. *Computers & Mathematics with Applications*, 70(8), 1921–1947. doi:10.1016/J.CAMWA. 2015.08.002.
- Delis, A. I., Nikolos, I. K., & Papageorgiou, M. (2016). Simulation of the penetration rate effects of ACC and CACC on macroscopic traffic dynamics. In *Proceedings of the IEEE 19th international conference on intelligent transportation systems (ITSC)* (pp. 336–341). IEEE. doi:10.1109/ITSC.2016.7795576.
- Delis, Anargiros, I., Nikolos, I. K., & Papageorgiou, M. (2018). A macroscopic multilane traffic flow model for ACC/CACC traffic dynamics. *Transportation Research Record*, 2672(20), 178–192. doi:10.1177/0361198118786823.
- Deng, W., Lei, H., & Zhou, X. (2013). Traffic state estimation and uncertainty quantification based on heterogeneous data sources: A three detector approach. *Transportation Research Part B Methodological*, 57, 132–157. doi:10.1016/j.trb.2013.08. 015.
- Diakaki, C., Papageorgiou, M., Papamichail, I., & Nikolos, I. (2015). Overview and analysis of vehicle automation and communication systems from a motorway traffic management perspective. *Transportation Research Part A Policy and Practice*, 75, 147–165. doi:10.1016/j.tra.2015.03.015.
- Diakaki, C., Papageorgiou, M., Papamichail, I., Nikolos, I., Iordanidou, G.-.R., & Porfyri, K. (2014). Overview and analysis of vehicle automation and communication systems from a motorway traffic management perspective. Retrieved from http: //www.traman21.tuc.gr/docs/wp1/TRAMAN21-D1-v4.pdf.
- Erlien, S. (2015). Shared vehicle control using safe driving envelopes for obstacle avoidance and stability (Ph.D. Thesis). Stanford University.
- Ferrara, A., Sacone, S., & Siri, S. (2015). Event-triggered model predictive schemes for freeway traffic control. *Transportation Research Part C Emerging Technologies*, 58, 554–567. doi:10.1016/j.trc.2015.01.020.
- Fountoulakis, M., Bekiaris-Liberis, N., Roncoli, C., Papamichail, I., & Papageorgiou, M. (2017). Highway traffic state estimation with mixed connected and conventional vehicles: Microscopic simulation-based testing. *Transportation Research Part C Emerging Technologies*, 78, 13–33. doi:10.1016/j.trc.2017.02.015.
- Gao, Y., Gray, A., Tseng, H. E., & Borrelli, F. (2014). A tube-based robust nonlinear predictive control approach to semiautonomous ground vehicles. *Vehicle System Dynamics*, 52(6), 802–823. doi:10.1080/00423114.2014.902537.
- Gao, Y., Lin, T., Borrelli, F., Tseng, E., & Hrovat, D. (2010). Predictive control of autonomous ground vehicles with obstacle avoidance on slippery roads. Retrieved from doi:10.1115/DSCC2010-4263.
- Gipps, P. G. (1981). A behavioural car-following model for computer simulation. Transportation Research Part B, 15(2), 105–111. doi:10.1016/0191-2615(81) 90037-0.
- Gray, A., Ali, M., Gao, Y., Hedrick, J. K., & Borrelli, F. (2012). Semi-autonomous vehicle control for road departure and obstacle avoidance. In Proceedings of the American control conference (ACC) (pp. 4239–4244). IEEE.

- Guériau, M., Billot, R., Faouzi, N.-E. E., Hassas, S., & Armetta, F. (2015). X2V-based information dissemination for highway congestion reduction. In *Proceedings of* the 18th Euro working group on transportation. EWGT.
- Gunter, G., Gloudemans, D., Stern, R.E., McQuade, S., Bhadani, R., Bunting, M. et al. (2019). Are commercially implemented adaptive cruise control systems string stable? Retrieved from https://arxiv.org/abs/1905.02108.
- Gunter, G., Janssen, C., Barbour, W., Stern, R.E., & Work, D.B. (2019). Model based string stability of adaptive cruise control systems using field data. Retrieved from http://arxiv.org/abs/1902.04983.
- Hall, R. W., & Caliskan, C. (1999). Design and evaluation of an automated highway system with optimized lane assignment. *Transportation Research Part C Emerging Technologies*, 7(1), 1–15. doi:10.1016/S0968-090X(99)00006-6.
- Hall, R. W., & Lotspeich, D. (1996). Optimized lane assignment on an automated highway. Transportation Research Part C Emerging Technologies, 4(4), 211–229. doi:10.1016/S0968-090X(96)00012-5.
- Herrera, J. C., & Bayen, A. M. (2010). Incorporation of Lagrangian measurements in freeway traffic state estimation. *Transportation Research Part B Methodological*, 44(4), 460–481. doi:10.1016/j.trb.2009.10.005.
- Hoogendoorn, S. P., & Bovy, P. H. L. (2001). State-of-the-art of vehicular traffic flow modelling. Proceedings of the Institution of Mechanical Engineers. Part I: Journal of Systems and Control Engineering.. doi:10.1243/0959651011541120.
- Ioannou, P. A., & Stefanovic, M. (2005). Evaluation of acc vehicles in mixed traffic: Lane change effects and sensitivity analysis. *IEEE Transactions on Intelligent Transportation Systems*, 6(1), 79–89. doi:10.1109/TITS.2005.844226.
 Ji, J., Khajepour, A., Melek, W. W., & Huang, Y. (2017). Path planning and tracking for
- Ji, J., Khajepour, A., Melek, W. W., & Huang, Y. (2017). Path planning and tracking for vehicle collision avoidance based on model predictive control with multiconstraints. *IEEE Transactions on Vehicular Technology*, 66(2), 952–964. doi:10.1109/ TVT.2016.2555853.
- Karafyllis, I., Bekiaris-Liberis, N., & Papageorgiou, M. (2018). Feedback control of nonlinear hyperbolic PDE systems inspired by traffic flow models. *IEEE Trans*actions on Automatic Control. doi:10.1109/TAC.2018.2887141.
- Karafyllis, I., Kontorinaki, M., & Papageorgiou, M. (2017). Robust global adaptive exponential stabilization of discrete-time systems with application to freeway traffic control. *IEEE Transactions on Automatic Control*, 62(12), 6195–6208. doi:10.1109/TAC.2017.2699125.
- Karafyllis, I., & Papageorgiou, M. (2019). Feedback control of scalar conservation laws with application to density control in freeways by means of variable speed limits. *Automatica*, 105, 228–236. doi:10.1016/j.automatica.2019.03.021.
- Kesting, A., Treiber, M., Schönhof, M., & Helbing, D. (2008). Adaptive cruise control design for active congestion avoidance. *Transportation Research Part C Emerging Technologies*, 16(6), 668–683. doi:10.1016/j.trc.2007.12.004.
- Kim, K., Cho, D., II, & Medanić, J. V. (2005). Lane assignment using a genetic algorithm in the automated highway systems. In *Proceedings of the IEEE conference* on intelligent transportation systems, *ITSC: 2005* (pp. 332–337). doi:10.1109/ITSC. 2005.1520070.
- Kontorinaki, M., Karafyllis, I., & Papageorgiou, M. (2019). Global exponential stabilisation of acyclic traffic networks. *International Journal of Control*, 92(3), 564–584. doi:10.1080/00207179.2017.1362114.
- Kontorinaki, M., Spiliopoulou, A., Roncoli, C., & Papageorgiou, M. (2017). First-order traffic flow models incorporating capacity drop: Overview and real-data validation. Transportation Research Part B Methodological, 106, 52–75. doi:10.1016/j.trb. 2017.10.014.
- Koren, Y., & Borenstein, J. (1991). Potential field methods and their inherent limitations for mobile robot navigation. In *Proceedings of the IEEE international conference on robotics and automation proceedings* (pp. 1398–1404). Institute of Electrical and Electronics Engineers (IEEE). doi:10.1109/robot.1991.131810.
- Kotsialos, A., & Papageorgiou, M. (2001). The importance of traffic flow modeling for motorway traffic control. *Networks and Spatial Economics*, 1(1–2), 179–203. doi:10.1023/A:1011537329508.
- Lee, J.-. K., & Lee, J.-. J. (1997). Discrete event modeling and simulation for flow control in an automated highway system. *Transportation Research Part C Emerging Technologies*, 5(3-4), 179–195. doi:10.1016/S0968-090X(97)00010-7.
- Lewis, F. L., Vrabie, D. L., & Syrmos, V. L. (2012). Optimal control (Third ed.). Hoboken, New Jersey: John Wiley & Sons, Inc. doi:10.1002/9781118122631.
- Li, P. Y., Horowitz, R., Alvarez, L., Frankel, J., & Robertson, A. M. (1997). An automated highway system link layer controller for traffic flow stabilization. *Transportation Research Part C Emerging Technologies*, 5(1), 11–37. doi:10.1016/S0968-090X(96) 00024-1.
- Li, Z., Liu, P., Xu, C., & Wang, W. (2016). Optimal mainline variable speed limit control to improve safety on large-scale freeway segments. *Computer-Aided Civil and Infrastructure Engineering*, 31(5), 366–380. doi:10.1111/mice.12164.
- Lipp, T., & Boyd, S. (2016). Variations and extension of the convex-concave procedure. Optimization and Engineering. doi:10.1007/s11081-015-9294-x.
- Lopez, P. A., Behrisch, M., Bieker-Walz, L., Erdmann, J., Flotterod, Y.-. P., Hilbrich, R., et al. (2018). Microscopic traffic simulation using SUMO. In Proceedings of the 21st international conference on intelligent transportation systems (ITSC) (pp. 2575–2582). IEEE. doi:10.1109/ITSC.2018.8569938.
- Makantasis, K., & Papageorgiou, M. (2018). Motorway path planning for automated road vehicles based on optimal control methods. *Transportation Research Record Journal of the Transportation Research Board*, 2672(19), 112–123. doi:10.1177/ 0361198118780710.
- Manolis, D., Papamichail, I., Kosmatopoulos, E. B., & Papageorgiou, M. (2016). Automated tuning of its management and control systems: Results from real-life experiments. *Transportation Research Part C Emerging Technologies*, 66. doi:10.1016/ j.trc.2015.08.020.

- Markantonakis, V., Skoufoulas, D. I., Papamichail, I., & Papageorgiou, M. (2019). Integrated traffic control for freeways using variable speed limits and lane change control actions. *Transportation Research Record* 0361198119846476. doi:10.1177/ 0361198119846476.
- Milanés, V., & Shladover, S. E. (2014). Modeling cooperative and autonomous adaptive cruise control dynamic responses using experimental data. *Transportation Research Part C Emerging Technologies*, 48, 285–300. doi:10.1016/j.trc.2014.09. 001.
- Montanino, M., & Punzo, V. (2013). Making NGSIM data usable for studies on traffic flow theory. Transportation Research Record Journal of the Transportation Research Board, 2390, 99–111. doi:10.3141/2390-11.
- Muralidharan, A., & Horowitz, R. (2015). Computationally efficient model predictive control of freeway networks. *Transportation Research Part C Emerging Technolo*gies, 58, 532–553. doi:10.1016/j.trc.2015.03.029.
- Ngoduy, D. (2012). Application of gas-kinetic theory to modelling mixed traffic of manual and ACC vehicles. *Transportmetrica*, 8(1), 43–60. doi:10.1080/ 18128600903578843.
- Ngoduy, D. (2013). Instability of cooperative adaptive cruise control traffic flow: A macroscopic approach. *Communications in Nonlinear Science and Numerical Simulation*, 18(10), 2838–2851. doi:10.1016/j.cnsns.2013.02.007.
- Ngoduy, D., Hoogendoorn, S. P., & Liu, R. (2009). Continuum modeling of cooperative traffic flow dynamics. *Physica A Statistical Mechanics and Its Applications*, 388(13), 2705–2716. doi:10.1016/j.physa.2009.02.040.
- Ngoduy, D., & Jia, D. (2017). Multi anticipative bidirectional macroscopic traffic model considering cooperative driving strategy. *Transportmetrica B*, 5(1), 100– 114. doi:10.1080/21680566.2016.1142401.
- Nikolos, I. K., Delis, A. I., & Papageorgiou, M. (2015). Macroscopic modelling and simulation of acc and cacc traffic. In Proceedings of the IEEE 18th international conference on intelligent transportation systems (pp. 2129–2134). IEEE. doi:10. 1109/ITSC.2015.344.
- Ntousakis, I. A., Nikolos, I. K., & Papageorgiou, M. (2015). On microscopic modelling of adaptive cruise control systems. *Transportation Research Procedia*, 6, 111–117. doi:10.1016/j.trpro.2015.03.010.
- Ntousakis, I. A., Nikolos, I. K., & Papageorgiou, M. (2016). Optimal vehicle trajectory planning in the context of cooperative merging on highways. *Transportation Research Part C Emerging Technologies*, 71, 464–488. doi:10.1016/j.trc.2016.08.007.
- Ntousakis, I.A., .Porfyri, K., Nikolos, I.K., .& Papageorgiou, M. (2014). Assessing the impact of a cooperative merging system on highway traffic using a microscopic flow simulator. In Volume 12: Transportation Systems. doi:10.1115/ IMECE2014-39850.
- Papadopoulou, S., Roncoli, C., Bekiaris-Liberis, N., Papamichail, I., & Papageorgiou, M. (2018). Microscopic simulation-based validation of a per-lane traffic state estimation scheme for highways with connected vehicles. *Transportation Research Part C Emerging Technologies*, 86, 441–452. doi:10.1016/j.trc.2017.11.012.
- Papageorgiou, M., Marinaki, M., Typaldos, P., & Makantasis, K. (2016). A feasible direction algorithm for the numerical solution of optimal control problems - extended version.
- Papageorgiou, M., Papamichail, I., Messmer, A., & Wang, Y. (2010). Traffic simulation with METANET. In J. Barceló (Ed.), Fundamentals of traffic simulation (pp. 399– 430). New York, NYNew York: Springer. doi:10.1007/978-1-4419-6142-6_11.
- Perraki, G., Roncoli, C., Papamichail, I., & Papageorgiou, M. (2018). Evaluation of a model predictive control framework for motorway traffic involving conventional and automated vehicles. *Transportation Research Part C Emerging Technologies*, 92, 456–471. doi:10.1016/j.trc.2018.05.002.
- Piccoli, B., Han, K., Friesz, T. L., Yao, T., & Tang, J. (2015). Second-order models and traffic data from mobile sensors. *Transportation Research Part C Emerging Tech*nologies, 52, 32–56. doi:10.1016/j.trc.2014.12.013.
- Porfyri, K. N., Nikolos, I. K., Delis, A. I., & Papageorgiou, M. (2016). Calibration of a second-order traffic flow model using a metamodel-assisted differential evolution algorithm. In Proceedings of the IEEE 19th international conference on intelligent transportation systems (ITSC) (pp. 366–371). IEEE. doi:10.1109/ITSC.2016. 7795581.
- Porfyri, K. N., Delis, A. I., Nikolos, I. K., & Papageorgiou, M. (2017). Calibration and validation of a macroscopic multi-lane traffic flow model using a differential evolution algorithm. In *Proceedings of the transportation research board 96th annual meeting* (pp. 17–01340). Retrieved from.https://trid.trb.org/view/1437576.
- Ramaswamy, D., Medanic, J. V., Perkins, W. R., & Benekohal, R. F. (1997). Lane assignment on automated highway systems. *IEEE Transactions on Vehicular Technology*, 46(3), 755–769. doi:10.1109/25.618201.
- Rao, B. S. Y., & Varaiya, P. (1994). Roadside intelligence for flow control in an intelligent vehicle and highway system. *Transportation Research Part C*, 2(1), 49–72. doi:10.1016/0968-090X(94)90019-1.
- Rasekhipour, Y., Khajepour, A., Chen, S. K., & Litkouhi, B. (2017). A potential fieldbased model predictive path-planning controller for autonomous road vehicles. *IEEE Transactions on Intelligent Transportation Systems*, 18(5), 1255–1267. doi:10. 1109/TITS.2016.2604240.
- Rios-Torres, J., & Malikopoulos, A. A. (2017). A survey on the coordination of connected and automated vehicles at intersections and merging at highway onramps. *IEEE Transactions on Intelligent Transportation Systems*, 18(5), 1066–1077. doi:10.1109/ITTS.2016.2600504.
- Roncoli, C., Papamichail, I., & Papageorgiou, M. (2015). Model predictive control for motorway traffic with mixed manual and VACS-equipped vehicles. *Transportation Research Procedia*: 10. doi:10.1016/j.trpro.2015.09.095.
- Roncoli, C., Bekiaris-Liberis, N., & Papageorgiou, M. (2016a). Optimal lane-changing control at motorway bottlenecks. In Proceedings of the IEEE 19th international

conference on intelligent transportation systems (ITSC) (pp. 1785-1791). IEEE. doi:10.1109/ITSC.2016.7795800.

- Roncoli, C., Bekiaris-Liberis, N., & Papageorgiou, M. (2016b). Use of speed measurements for highway traffic state estimation: Case studies on NGSIM data and highway A20, Netherlands. *Transportation Research Record Journal of the Transportation Research Board*, 2559, 90–100. doi:10.3141/2559-11.
- Roncoli, C., Bekiaris-Liberis, N., & Papageorgiou, M. (2017). Lane-changing feedback control for efficient lane assignment at motorway bottlenecks. *Transportation Research Record Journal of the Transportation Research Board*, 2625, 20–31. doi:10.3141/2625-03.
- Roncoli, C., Papageorgiou, M., & Papamichail, I. (2015a). Motorway traffic flow optimisation in presence of vehicle automation and communication systems. *Computational Methods in Applied Sciences*, 38, 1–16. doi:10.1007/ 978-3-319-18320-6_1.
- Roncoli, C., Papageorgiou, M., & Papamichail, I. (2015b). Traffic flow optimisation in presence of vehicle automation and communication systems – part I: A first-order multi-lane model for motorway traffic. *Transportation Research Part C Emerging Technologies*, 57, 241–259. doi:10.1016/J.TRC.2015.06.014.
- Roncoli, C., Papageorgiou, M., & Papamichail, I. (2015c). Traffic flow optimisation in presence of vehicle automation and communication systems – part II: Optimal control for multi-lane motorways. *Transportation Research Part C Emerging Tech*nologies, 57, 260–275. doi:10.1016/J.TRC.2015.05.011.
- Schakel, W. J., & van Arem, B. (2014). Improving traffic flow efficiency by in-car advice on lane, speed, and headway. *IEEE Transactions on Intelligent Transportation Systems*, 15(4), 1597–1606. doi:10.1109/TITS.2014.2303577.
- Schakel, W. J., Van Arem, B., & Netten, B. D. (2010). Effects of cooperative adaptive cruise control on traffic flow stability. In *Proceedings of the IEEE Conference* on Intelligent Transportation Systems, ITSC (pp. 759–764). doi:10.1109/ITSC.2010. 5625133.
- Schmidt, G., & Posch, B. (2010). Automatische zusammmenführung zweier fahrzeugströme – Eine rückblende. At - Automatisierungstechnik Methoden Und Anwendungen Der Steuerungs-, Regelungs- Und Informationstechnik. doi:10.1524/ auto.2010.0842.
- Seo, T., Kusakabe, T., & Asakura, Y. (2015). Estimation of flow and density using probe vehicles with spacing measurement equipment. *Transportation Research Part C Emerging Technologies*, 53, 134–150. doi:10.1016/j.trc.2015.01.033.
- Shladover, S. E., Su, D., & Lu, X.-, Y. (2012). Impacts of cooperative adaptive cruise control on freeway traffic flow. *Transportation Research Record Journal of the Transportation Research Board*, 2324(1), 63–70. doi:10.3141/2324-08.
- Shum, A., Morris, K., & Khajepour, A. (2015). Direction-dependent optimal path planning for autonomous vehicles. *Robotics and Autonomous Systems*, 70, 202–214. doi:10.1016/j.robot.2015.02.003.
- Shvetsov, V., & Helbing, D. (1999). Macroscopic dynamics of multilane traffic. *Physical Review E*, 59(6), 6328–6339. doi:10.1103/PhysRevE.59.6328.
- Spiliopoulou, A., Manolis, D., Vandorou, F., & Papageorgiou, M. (2018). Adaptive cruise control operation for improved motorway traffic flow. *Transportation Research Record Journal of the Transportation Research Board*. doi:10.1177/ 0361198118796020.
- Strofylas, G. A., Porfyri, K. N., Nikolos, I. K., Delis, A. I., & Papageorgiou, M. (2018). Using synchronous and asynchronous parallel differential evolution for calibrating a second-order traffic flow model. *Advances in Engineering Software*, 125, 1– 18. doi:10.1016/j.advengsoft.2018.08.011.
- Tanzmeister, G., Friedl, M., Wollherr, D., & Buss, M. (2013). Path planning on grid maps with unknown goal poses. In Proceedings of the IEEE conference on intelligent transportation systems, ITSC (pp. 430–435). doi:10.1109/ITSC.2013. 6728269.
- Toledo, T., Ben-Akiva, M. E., Darda, D., Jha, M., & Koutsopoulos, H. N. (2007). Calibration of microscopic traffic simulation models with aggregate data. *Transportation Research Record Journal of the Transportation Research Board*, 1876(1), 10–19. doi:10.3141/1876-02.
- Torné, J., Soriguera, F., & Geroliminis, N. (2014). Coordinated active traffic management freeway strategies using capacity-lagged cell transmission model. In Proceedings of the transportation research board 93rd annual meeting (pp. 14-3941). Retrieved from.http://amoline.trb.org/trb-55856-2014a-1. 823612/t-1129-1.878850/369-1.882663/14-3941-1.882672/14-3941-1.882677/
- Treiber, M., Hennecke, A., & Helbing, D. (2000). Congested traffic states in empirical observations and microscopic simulations. *Physical Review E Statistical Physics Plasmas Fluids and Related Interdisciplinary Topics*, 62(2), 1805–1824. doi:10.1103/ PhysRevE.62.1805.
- Treiber, M., & Kesting, A. (2012). Validation of traffic flow models with respect to the spatiotemporal evolution of congested traffic patterns. *Transportation Research Part C Emerging Technologies*, 21(1), 31–41. doi:10.1016/j.trc.2011.09.002.
- Treiber, M., & Kesting, A. (2013). Traffic flow dynamics. Springer. doi:10.1007/ 978-3-642-32460-4.
- TSS-Transport Simulation Systems. (2014). Aimsun 8 users' manual. Retrieved from https://www.scribd.com/document/266245543/Aimsun-Users-Manual-v8.
- Typaldos, P., Mountakis, K., Papageorgiou, M., & Papamichail, I. (2019). Optimization-based path planning for automated vehicles. In Proceedings of the 9th international congress on transportation research.
- Typaldos, P., Papamichail, I., & Papageorgiou, M. (2018). Minimization of fuel consumption for vehicle trajectories. In Proceedings of the 97th annual meeting of the transportation research board (TRB) (pp. 18–03953). Retrieved from.https: //trid.trb.org/view/1496034.
- US DoT. (2005). Next generation SIMulation (NGSIM). Retrieved from www. ngsim-community.org.

- Varaiya, P. (1993). Smart cars on smart roads: Problems of control. IEEE Transactions on Automatic Control, 38(2), 195–207. doi:10.1109/9.250509.
- Wang, H., Huang, Y., Khajepour, A., Rasekhipour, Y., Zhang, Y., & Cao, D. (2019). Crash mitigation in motion planning for autonomous vehicles. *IEEE Transactions on Intelligent Transportation Systems*. doi:10.1109/TITS.2018.2873921.
- Wang, J., Wu, J., & Li, Y. (2015). The driving safety field based on driver-vehicle-road interactions. *IEEE Transactions on Intelligent Transportation Systems*, 16(4), 2303– 2314. doi:10.1109/TITS.2015.2401837.
- Wang, J., & Rajamani, R. (2004). Should adaptive cruise-control systems be designed to maintain a constant time gap between vehicles? *IEEE Transactions on Vehicular Technology*, 53(5), 1480–1490. doi:10.1109/TVT.2004.832386.
- Werling, M., & Liccardo, D. (2012). Automatic collision avoidance using modelpredictive online optimization. In Proceedings of the IEEE conference on decision and control (pp. 6309–6314). doi:10.1109/CDC.2012.6426612.
- Williams, R. L., & Lawrence, D. A. (2007). Linear state-space control systems. Hoboken, New Jersey: John Wiley & Sons, Inc. doi:10.1002/9780470117873.
- Wolf, M. T., & Burdick, J. W. (2008). Artificial potential functions for highway driving with collision avoidance. In *Proceedings of the IEEE international conference on robotics and automation* (pp. 3731–3736). doi:10.1109/ROBOT.2008.4543783.
- Yi, J., & Horowitz, R. (2006). Macroscopic traffic flow propagation stability for adaptive cruise controlled vehicles. *Transportation Research Part C Emerging Technolo*gies, 14(2), 81–95. doi:10.1016/j.trc.2006.05.005.
- Yuan, Y., Van Lint, H., Van Wageningen-Kessels, F., & Hoogendoorn, S. (2014). Network-wide traffic state estimation using loop detector and floating car data. *Journal of Intelligent Transportation Systems Technology Planning and Operations*, 18, 41–50 Taylor and Francis Inc. doi:10.1080/15472450.2013.773225.
- Zhang, Y., & Ioannou, P. A. (2017). Combined variable speed limit and lane change control for highway traffic. *IEEE Transactions on Intelligent Transportation Systems*. 18(7), 1812–1823. doi:10.1109/TITS.2016.2616493.
- Zheng, L., Jin, P. J., & Huang, H. (2015). An anisotropic continuum model considering bi-directional information impact. *Transportation Research Part B Methodological*, 75, 36–57. doi:10.1016/j.trb.2015.02.011.
- Ziegler, J., Bender, P., Dang, T., & Stiller, C. (2014). Trajectory planning for bertha – a local, continuous method. In *Proceedings of the IEEE intelligent vehicles* symposium (pp. 450–457). Institute of Electrical and Electronics Engineers Inc. doi:10.1109/IVS.2014.6856581.



Ioannis Papamichail is an Associate Professor with the School of Production Engineering & Management, Technical University of Crete, Greece. He is the author and a coauthor of several technical papers in scientific journals and conference proceedings. His main research interests include automatic control and optimization theory and applications to traffic and transportation systems. Dr. Papamichail is an Associate Editor for IEEE Transactions on Intelligent Transportation Systems and a Member of the Editorial Advisory Board for Transportation Research Part C: Emerging Technologies. He received the 1998 Eugenidi Foundation Scholarship for Postgraduate Studies and the 2010 Transition to Practice Award from the IEEE Control

Systems Society.



Nikolaos Bekiaris-Liberis is an Assistant Professor with the School of Electrical & Computer Engineering and Marie Sklodowska-Curie Fellow with the School of Production Engineering & Management, Technical University of Crete, Greece. He received the Ph.D. degree in Aerospace Engineering from the University of California, San Diego, in 2013. From 2013 to 2014 he was a postdoctoral researcher at the University of California, Berkeley and from 2014 to 2017 he was a research associate and adjunct professor at Technical University of Crete, Greece. He has co-authored the SIAM book Nonlinear Control under Nonconstant Delays. His interests are in delay systems, distributed parameter systems, nonlinear control,

and their applications. Dr. Bekiaris-Liberis received the Chancellor's Dissertation Medal in Engineering from UC San Diego, in 2014 and is the recipient of a 2016 Marie Sklodowska-Curie Individual Fellowship Grant.



Anargiros I. Delis is an Associate Professor with the School of Production Engineering & Management, Technical University of Crete, Greece and a collaborative member at FORTH-IACM. He holds a Mathematics degree from the University of Crete, Greece, an M.Sc. in Numerical Analysis and Computing from Manchester University and UMIST, U.K., and received his Ph.D. in Computational and Applied Mathematics from the University of West of England, Bristol in 1998. His research interests are focused on the numerical solution of PDE's, Computational Fluid Dynamics and Scientific Computing with emphasis to environmental and geophysical flows.



Diamantis Manolis received the Diploma degree in Production Engineering and Management, Technical University of Crete, Greece, in 2003 and the M.Sc. degree in Operation Research in 2006. He also received the Ph.D. degree in Operation Research from the Technical University of Crete in 2010. From 2004 he is a research associate with the Dynamic Systems and Simulation Laboratory, Technical University of Crete, Greece. His main research interests include automatic control and optimization theory and applications to traffic and transportation systems.



Kiriakos-Simon Mountakis graduated from the Department of Electronic & Computer Engineering, Technical University of Crete. He then obtained an MSc and PhD in Computer Science from the Electrical Engineering, Mathematics and Computer Science faculty, Delft University of Technology), as a member of the Algorithmics group. His PhD dissertation, entitled "Scheduling Under Uncertainty: Attaining Flexibility, Robustness and Stability", was part of the Rolling-Stock Life-Cycle Logistics applied R&D program by the Dutch railways and focused on scheduling approaches that mitigate the effects of uncertainty. Since 2017, he has been a Research Associate with the Dynamic Systems Simulation Laboratory, Technical Uni-

versity of Crete, focusing on various control methods for self-driving cars. From a broad perspective, his research interests include various combinatorial optimization problems in Artificial Intelligence and Operational Research with an emphasis on Simple Temporal Problems and (stochastic) Resource-Constrained Scheduling; various network-flow problems; mathematical programming.



Ioannis K. Nikolos, Mechanical Engineer, is a Professor with the School of Production Engineering & Management, Technical University of Crete, Greece. He has more than 25 years of experience in R&D projects funded by the EU, the Industry, and the Greek State. His research work is in the fields of fluid mechanics, CFD, computational engineering, and traffic flow modelling. He has coauthored three books, 54 journal papers and book chapters, and more than 100 conference papers. His work has attracted more than 1700 citations. He is a member of the ASME, AIAA, IEEE, and the Technical Chamber of Greece.



Claudio Roncoli is an Assistant Professor of Transportation Engineering at Aalto University, Finland. He completed his PhD degree in System monitoring and environmental risk management (2013) at University of Genova, Italy.

Before joining Aalto University, he was a research assistant at University of Genova, Italyö a visiting research assistant at Imperial College London, UK and a Postdoctoral Research Associate at Technical University of Crete, Greece.

Dr. Roncoli has been involved in several national and international research projects, also as principal investigator. His research interests include real-time traffic man-

agement; modelling, optimisation, and control of traffic systems with connected and automated vehicles, as well as smart mobility and intelligent transportation systems.



Markos Papageorgiou was a Professor of Automation at the Technical University of Munich (1988–1994). Since 1994 he has been a Professor at the Technical University of Crete, Greece. He was a Visiting Professor at Politecnico di Milano, Ecole Nationale des Ponts et Chaussées, MIT, University of Rome La Sapienza and Tsinghua University; and a Visiting Scholar at the UCBerkeley. His research interests include automatic control theory and applications to traffic and transportation systems, water systems and further areas. He is Life Fellow of IEEE and Fellow of IFAC. He received several distinctions and awards, including an ERC Advanced Investigator Grant.